

ONTARIO

SUPERIOR COURT OF JUSTICE

WEDNESDAY, THE 28th

DAY OF MARCH, 2012

BETWEEN:

MARVIN SHEREBRIN and MICHAEL KRAUSE

Plaintiffs

- and -

BUILDING PRODUCTS OF CANADA CORP.

Defendant

Proceeding under the Class Proceedings Act, 1992

ORDER

THIS MOTION made by Class Counsel for an Order certifying this action as a class proceeding for settlement purposes and approving notice in the form attached as Schedule 'B' was heard on March 28, 2012 at the Court House in London, Ontario.

ON READING the materials filed by the Plaintiffs; and

ON HEARING the submissions of counsel for the Plaintiffs and the Defendant:

- 1. **THIS COURT ORDERS** that for the purposes of this Order, the definitions set out in the Agreement annexed as Schedule "A" apply to and are incorporated into the Order.
- 2. **THIS COURT ORDERS** that the within proceeding is certified as a class action for settlement purposes only, and subject to the terms and conditions of the Agreement and the conditions set out herein.
- THIS COURT ORDERS that the certified class is defined as follows:
 - "All individuals and entities that own or owned homes, residences, buildings, or other structures located in Canada whose roofs contain or contained BP Organic Shingles, excluding members of the Quebec Class."
- 4. **THIS COURT ORDERS** that Marvin Sherebrin and Michael Krause be appointed as representative plaintiffs on behalf of the Class.
- 5. **THIS COURT ORDERS** that this action is certified for settlement purposes on the basis of the following common issue:
 - i. Is the Defendant liable to the Settlement Class Members in negligence, breach of statute or warranty?
- 6. **THIS COURT ORDERS** that the following persons be excluded from the Canadian National Class:
 - All persons who timely exercise their rights under section 9 of the Class Proceedings Act, 1992, S.O. 1992 c. 6 (the "CPA") pursuant to the terms of the Agreement and the Order;
 - ii. All persons who filed a claim concerning their BP Organic Shingles in any court of law and whose claim has been resolved with a final judgment or settlement, whether or not favourable to the person; and
 - iii. The Defendant, any entity in which the Defendant has a controlling interest, any entity that has a controlling interest in the Defendant, and the Defendant's legal representatives, assigns and successors.

- 7. THIS COURT ORDERS that a putative Settlement Class Member may opt out of the Canadian National Class by sending a fully completed Opt Out Form to the office of Class Counsel, at an address to be identified in the Settlement Notice, postmarked no later than 60 days following the first publication of the Settlement Notice.
- 8. THIS COURT ORDERS that any Class Member who has validly opted out of this action:
 - a. shall be excluded from the Canadian National Class;
 - shall have no rights and be entitled to no relief under this or any subsequent orders in the action or under the Agreement (including no payments as provided in the Agreement); and
 - c. shall not be bound by the Agreement, by the Order, or the Approval Order, if granted.
- 9. THIS COURT ORDERS that any putative Settlement Class Members who have opted out of the Canadian National Class may withdraw their request to opt out prior to the Effective Date, but only if they accept the benefits and terms of the Agreement and dismiss with prejudice any other pending action against the Defendants arising from damage to their affected buildings, homes, residences, or other structures caused by any alleged defects in the BP Organic Shingles installed in those properties.
- 10. THIS COURT ORDERS that any putative Settlement Class Member who does not submit a written request to opt out within the time prescribed, in accordance with the Agreement and the terms of the Order, so as to validly opt-out of the Canadian National Class, shall be automatically included in the Canadian National Class and shall be bound by the terms of the Agreement, the Order and by all subsequent proceedings, orders and judgments in the action.
- 11. THIS COURT ORDERS that Settlement Class Members may object to the Agreement by filing a written objection to the office of Class Counsel, at an address to be identified in the Settlement Notice, postmarked no later than 60 days following the first publication of the Settlement Notice.

12. **THIS COURT ORDERS** that:

- a. the written objection must bear the signature of the Settlement Class Member;
- b. if the Settlement Class Member is represented by counsel, the written objection must also bear the signature of counsel; and
- c. the written objection must specify the following information:
 - i. the Settlement Class Member's current address and telephone number;
 - ii. the address of the property(ies) that may contain BP Organic Shingles;
 - iii. the number of units of residential property or other structures at each address owned by the Settlement Class Member containing BP Organic Shingles;
 - iv. the exact nature of the objection; and
 - v. whether or not the Settlement Class Member intends to appear at the Final Approval Hearing.
- 13. **THIS COURT ORDERS** that Class Counsel shall promptly furnish counsel for the Defendant with copies of any written objections that come into their possession.
- 14. THIS COURT ORDERS that no person shall be entitled to contest the approval of the terms and conditions of the Agreement or the Approval Order requested, except by filing and serving a written objection (and then appearing at the Final Approval Hearing), in accordance with the provisions of this Order and the Agreement. Any Settlement Class Member who does not submit a timely, written objection in compliance with all of the procedures set forth in the Order and the Agreement shall be deemed to and shall have waived any and all objections, and will therefore be bound by all proceedings, orders, and judgments in this case which will be preclusive in all pending or future lawsuits or other proceedings.

- 15. **THIS COURT ORDERS** that if the Agreement is terminated for any reason or if any specified condition to the Agreement is not satisfied and any party seeks to terminate the Agreement, or if the Agreement is not approved for any reason whatsoever by this Court, or any appellate court, then:
 - this Order shall be set aside, be of no further force or effect, and be without prejudice to the positions or rights of any party;
 - ii. the Agreement and all proceedings in connection therewith shall be null and void, except insofar as expressly provided in the Agreement and without prejudice to the *status quo ante* rights of the Plaintiffs and the Defendant;
 - this class action in respect of the Canadian National Class shall immediately be decertified as a class proceeding pursuant to section 10 of the CPA without prejudice to the Plaintiffs' ability to reapply for certification;
 - iv. each party to the action shall be restored to his, her, or its respective position as it existed immediately prior to the execution of the Agreement;
 - v. the Agreement and the fact that it was entered into shall not be offered, received, or construed as an admission or as evidence for any purpose in this or any other proceeding, including, without limitation, the ability of any other class to be certified; and
 - vi. the Agreement, actions in conformance with or in furtherance of the Agreement, and the other documents prepared or executed by any Party in negotiating or implementing the Agreement, including any terms of any such documents, shall never be offered in evidence in or shared with any

party to any civil, criminal, or administrative action or proceeding without the Defendant's express written consent.

- 16. **THIS COURT ORDERS** that the entry of this Order is without prejudice to the rights of the Defendant to terminate or void the Agreement in accordance with its terms.
- 17. THIS COURT ORDERS the summary form and long form of the Settlement Notice to Settlement Class Members is approved substantially in the forms attached hereto as Schedule 'B'.
- 18. **THIS COURT ORDERS** that the method of dissemination of the Settlement Notice is approved substantially in the form attached hereto as Schedule 'C' and the Settlement Notices shall be disseminated in accordance with the method of dissemination, the Agreement and the Order.
- 19. THIS COURT ORDERS that NPT RicePoint Class Action Services is hereby appointed to disseminate the Settlement Notices in accordance with the method of dissemination, the Agreement, and the terms of this Order and that NPT Ricepoint Class Action Services has consented to same.
- 20. **THIS COURT ORDERS** that Counsel for the Parties will jointly oversee the implementation of the method of dissemination by NPT RicePoint Class Action Services.
- 21. **THIS COURT ORDERS** that the forms and manner of notice as set out in the Order and the method of dissemination are sufficient and satisfy the requirements of notice under section 19 of the CPA.
- 22. **THIS COURT ORDERS** that the Opt Out Form is approved substantially in the form attached hereto as Schedule 'D'.
- 23. THIS COURT ORDERS that the Claim Form is approved substantially in the form attached hereto as Schedule 'E'.

- 24. THIS COURT ORDERS that, notwithstanding section 9.7 of the Agreement, the Settlement Website shall include a form that Class Members can complete to request a copy of the Claim Form. Where a Class Member has completed the form, the Defendants shall send a copy of the Claim Form by email, if an email address has been provided, or by regular mail to the Class Member within two business days.
- 25. THIS COURT ORDERS that a Settlement Approval Hearing shall be held on 2012 at 10:00am, at the London Court House, 80 Dundas Street, London, Ontario.
- 26. **THIS COURT ORDERS** that the date and time of the Final Approval Hearing shall be published in the Settlement Notices but may be adjourned by the Court without further notice to Settlement Class Members other than that which may be posted at the Court and/or on the Settlement Website.
- 27. **THIS COURT ORDERS** that this Order shall be of no force and effect until equivalent Orders are obtained in the United States and Quebec.

Madam Justice Leitch

Ontario Superior Court of Justice

ORDER ENTERED

MAR 2 8 2012

524

Marvin Sherebrin et al. v Building Products of Canada Corp. Diane Fitzsimmons v La Cie Materiaux de Construction BP Canada Robert S. Melillo et al. v Building Products of Canada Corp. Ontario Superior Court of Justice Court File No. 4367/11CP Superior Court of Quebec Court File No. 500-06-00580-114 Vermont Superior Court, Chittenden Unit, Court File No. s618-11 cnc

AGREEMENT OF COMPROMISE AND SETTLEMENT

1.	RECITALS	2
2.	DEFINITIONS	4
3.	CLASS CERTIFICATION	10
4.	CLAIMS TO BE COMPENSATED BY THE AGREEMENT	11
5.	CLAIMS PROCESS	16
6.	SETTLEMENT BENEFITS	31
7.	CLASS COUNSEL FEES AND EXPENSES	36
8.	PRELIMINARY APPROVAL ORDERS	36
9.	NOTICE OF SETTLEMENT	37
10.	SETTLEMENT CLASS MEMBERS' RIGHT TO OPT OUT AND DISMISSAL OF OTHER PROCEEDINGS	40
11.	SETTLEMENT CLASS MEMBERS' RIGHT TO OBJECT	42
12.	FINAL APPROVAL ORDERS	43
13.	DEFENDANT'S RIGHT OF TERMINATION	44
14.	EFFECTIVE DATE	46
15.	RELEASE	46
16.	EXCLUSIVE REMEDY; DISMISSAL OF ACTION; JURISDICTION OF COURT	49
17	OTHER TERMS AND CONDITIONS	51

AGREEMENT OF COMPROMISE AND SETTLEMENT

The Plaintiffs, on behalf of themselves and the Settlement Classes set forth herein, and Building Products of Canada Corp./La Cie Materiaux de Construction BP Canada (the "Defendant"), stipulate and agree, pursuant to the terms and conditions set forth in this Agreement, to settle, dismiss, and compromise fully and finally the Actions.

1. <u>RECITALS</u>

- 1.1 WHEREAS the Actions have been commenced by the Plaintiffs in Ontario, Quebec and the United States which allege that the Defendant manufactured, distributed, marketed and/or sold BP Organic Shingles in Canada and the United States, which shingles were allegedly defective and prone to premature failure.
- 1.2 WHEREAS the Defendant denies all allegations of fault, wrongdoing, or liability made by the Plaintiffs and asserts that the BP Organic Shingles were sold without defects; and
- 1.3 WHEREAS the Parties have engaged in extensive, arms-length negotiations regarding the settlement of claims involving BP Organic Shingles; and
- 1.4 WHEREAS the Plaintiffs and Class Counsel have evaluated the time and expense that will be necessary to prosecute the Actions to final judgment, the delays that are likely before any judgment might be entered, and the uncertainty inherent in predicting the outcome of any complex litigation such as this and, based upon such evaluation, have concluded that further proceedings in the Actions are likely to be protracted, complex, and expensive, and that the outcome is uncertain; and
- 1.5 WHEREAS without conceding any lack of merit of any of their claims, Plaintiffs and Class Counsel have concluded that it is in the best interest of the Settlement Classes to settle

the Actions on the terms set forth herein, and that the settlement with the Defendant embodied in this Agreement is fair, reasonable, and adequate to Plaintiffs and the Settlement Classes; and

- 1.6 WHEREAS this Agreement and the settlement embodied herein shall in no event be construed as or deemed to be evidence of an admission or a concession on the part of the Defendant of any fault, wrongdoing, or liability whatsoever, or that any of the allegations in the Actions are true, and without conceding any infirmity in their defenses, the Defendant considers it desirable to enter into this Agreement in order to avoid further expense, dispose of what would be burdensome and protracted litigation, and avoid the uncertain outcome of proceeding with the Actions;
- 1.7 WHEREAS for the purposes of settlement only and contingent on approvals by the Courts as provided for in this Agreement, the Parties have consented to certification or authorization of the Actions as class proceedings and have consented to a Settlement Class in each of the Actions; and
- 1.8 WHEREAS the Plaintiffs assert that they are adequate class representatives for the Settlement Classes and will seek to be appointed representative plaintiffs in their respective Actions;

NOW, THEREFORE, it is hereby agreed by and between the Defendant and the Plaintiffs, acting for themselves and the Settlement Classes, by and through their respective attorneys, that, except as specifically stated to the contrary in this Agreement, all of the allegations, claims, demands, causes of action, and liabilities, which have been or could have been asserted by the Plaintiffs or Settlement Classes Members against any of the Released Parties relating to, arising out of, or in connection with any of the allegations made in the

Actions, shall be settled and compromised, and the Actions shall be dismissed with prejudice, according to the terms and conditions set forth below.

2. **DEFINITIONS**

As used in this Agreement, its exhibits, and the notices and other documents contemplated by this Agreement, and any amendments thereto, the following terms shall have the meanings set forth below. Terms used in the singular shall be deemed to include the plural and vice versa.

- 2.1 "Actions" shall mean the Canadian Actions and the U.S. Action.
- 2.2 "Adjudicator" shall mean an independent third-party agreed upon by the Parties and approved by the Courts to adjudicate Requests for Review in accordance with sections 5.25 to 5.33.
- 2.3 "Agreement" shall mean this Agreement of Compromise and Settlement, including the exhibits and notices and other documents contemplated by this Agreement, and any amendments thereto.
- 2.4 "Applicable Warranty" shall mean the relevant warranty for the particular BP Organic Shingles.
- 2.5 "Applicable Warranty Period" shall mean the relevant warranty period for the particular BP Organic Shingles, as set out in Exhibit "A".
 - 2.6 "Approved Repairs" shall have the meaning attributed to it in section 6.4(a).
- 2.7 "Approved Repairs Amount" shall have the meaning attributed to it in section 6.4(a)(i).
- 2.8 "BP Organic Shingles" shall mean shingles made with a felt reinforcement base material that is saturated with asphalt, also known as "organic" roofing shingles, manufactured

during the period from 1985 to 2010 under the brand names: Eclipse, Eclipse H/R, Eclipse LS, Super Eclipse, Weather-Tite, Mirage, Rampart, Tradition, Tite-Lok, Esgard Pro-Standard, Pro-Standard, Esgard 20, Esgard 25, Citadel, Tite-On, Roofmaster, Roofmaster Classic, Roofmaster Plus, Elegance, Elegance II, Europa, and Super Lok.

- 2.9 "Canadian Actions" shall mean the Ontario Action and the Quebec Action.
- 2.10 "Canadian National Class" shall have the meaning attributed to it in section 3.1.
- 2.11 "Canadian Class Representatives" shall mean those class representatives known as Marvin Sherebrin, Michael Krause, and Diane Fitzsimmons who are, collectively, the named Plaintiffs in the Canadian Actions, individually and on behalf of members of the Canadian National Class or the Quebec Class, as appropriate.
- 2.12 "Cash Settlement Option" shall mean the settlement benefit option described in section 6.5.
 - 2.13 "Causation Defense" shall have the meaning attributed to it in section 4.7.
- 2.14 "Claimant" shall mean a Settlement Class Member who submits a Claims Form under the terms of this Agreement.
- 2.15 "Claim Decision" shall mean the document used by the Defendant to advise Claimants of whether and to what extent their claims are approved or denied.
- 2.16 "Claim Form" shall mean the form agreed upon by the Parties, and approved by the Courts for use by Settlement Class Members in making claims under this Agreement.
- 2.17 "Class Counsel" shall mean counsel for the Plaintiffs in the Canadian Actions and the law firms appointed as lead counsel in the U.S. Action. For purposes of this Agreement, to the extent the duties of Class Counsel are enumerated, Class Counsel in the United States and Canada, respectively, shall only be obligated to act as counsel in their respective jurisdictions

and U.S. counsel shall not act as counsel for the Canadian Classes and Canadian counsel shall not act as counsel for the U.S. Class.

- 2.18 "Courts" shall mean the Ontario Court, the Quebec Court and the U.S. Court.
- 2.19 "Damage to BP Organic Shingles" shall have the meaning attributed to it in section 4.6.
- 2.20 "Damaged BP Organic Shingles" shall have the meaning attributed to it in section 4.6.
 - 2.21 "Defendant" shall have the meaning attributed to it in the preamble.
 - 2.22 "Effective Date" shall have the meaning attributed to it in section 14.2.
 - 2.23 "Eligible Claimant" shall have the meaning attributed to it in section 4.1.
- 2.24 "Final Approval Hearings" shall mean the hearings conducted by the Courts in connection with the determination of the fairness, adequacy, and reasonableness of this Agreement under section 29(2) of the *Class Proceedings Act*, 1992, S.O. 1992, c. 6, article 1025 of the Quebec *Code of Civil Procedure* and V.R.C.P. 23.
 - 2.25 "Final Approval Order" shall have the meaning attributed to it in section 12.1.
- 2.26 "Inspector" shall mean IRC Group and/or such other entity or entities as may be agreed upon by the Parties or appointed by the Courts to adjudicate Requests for Review in accordance with sections 5.25 to 5.33, disputes relating to the reasonableness of the Approved Repairs Amount and/or proration of the Approved Repairs Amount in accordance with section 6.4(a)(iv), and disputes relating to Repairs Invoices in accordance with section 6.4(h).
- 2.27 "Non-Proration Period" shall mean the later of: (i) the end of the period during which the Defendant is obligated to pay the costs of repair or replacement without any proration

pursuant to the terms of the Applicable Warranty; or (ii) five years from the installation of the BP Organic Shingles.

- 2.28 "Objection Deadline" shall mean the deadline set by the Courts for any Settlement Class Member to object to the Agreement.
- 2.29 "Ontario Action" shall mean the putative class action lawsuit pending in Ontario under the caption, Marvin Sherebrin et al. v. Building Products of Canada Corp., Ontario Superior Court of Justice Court File No. 4367/11CP.
 - 2.30 "Ontario Court" shall mean the Ontario Superior Court of Justice.
- 2.31 "Opt-Out Deadline" shall mean the deadline set by the Courts for any Settlement Class Member to opt out of the Settlement Class.
- 2.32 "Opt-Out Form" shall mean the form agreed upon by the Parties, and approved by the Courts for use by Settlement Class Members to opt out of the Settlement Class.
- 2.33 "Other Actions" shall mean actions or proceedings, other than the Actions, relating to claims released pursuant to section 15.1, commenced by a Settlement Class Member either before or after the Effective Date.
 - 2.34 "Parties" shall mean the Plaintiffs and the Defendant.
- 2.35 "Payment of Repairs Settlement Option" shall mean the settlement benefit option described in section 6.4.
 - 2.36 "Plaintiffs" shall mean the named plaintiffs in the Actions.
- 2.37 "Preliminary Approval Order" shall have the meaning attributed to it in section 8.1.
- 2.38 "Quebec Action" shall mean the proceeding commenced by motion to obtain authorization to exercise a class action and to obtain the status of representative under the style

of cause Diane Fitzsimmons v. La Cie Materiaux de Construction BP Canada, Quebec Superior Court File No. 500-06-00580-114.

- 2.39 "Quebec Class" shall have the meaning attributed to it in section 3.2.
- 2.40 "Quebec Court" shall mean the Superior Court of Quebec.
- 2.41 "Released Parties" shall have the meaning attributed to it in section 15.1.
- 2.42 "Releasing Parties" shall have the meaning attributed to it in section 15.1.
- 2.43 "Repair Dispute Form" shall mean the form agreed upon by the Parties, or as otherwise approved by the Courts, for use by Claimants to request a review of the Defendant's position that all or part of a Repairs Invoice submitted by the Claimant does not relate to Approved Repairs or that the Repairs Invoice does not establish that the Claimant used roofing shingles manufactured by the Defendant in completing any Approved Repairs.
- 2.44 "Repairs Invoice" shall mean any invoices, bills, or other documents relating to labour and material expenses incurred by a Claimant in the completion of Approved Repairs. Repairs Invoice shall not include any quotes or bids.
- 2.45 "Request for Review Form" shall mean the form agreed upon by the Parties, or as otherwise approved by the Courts for use by Claimants to request the review by the Inspector or the Adjudicator, as appropriate, of a Claim Decision issued by the Defendant in respect of their claims.
- 2.46 "Review Decision" shall mean the document used by the Adjudicator and the Inspector to advise Claimants and the Defendant of the determination made by the Adjudicator or the Inspector as to: (i) the correctness of the Claim Decision; (ii) whether the Claimant has acted in a manner that is fraudulent and/or abusive; (iii) to the extent the Review Decision is issued by the Inspector, whether the shingles at issue were in fact BP Organic Shingles; and (iv)

where the review relates to the reasonableness of the Approved Repairs Amount and/or proration of the Approved Repairs Amount, the appropriate Approved Repairs Amount and/or proration of the Approved Repairs Amount. The Review Decision shall contain a brief explanation of the basis for such determination.

- 2.47 "Review Materials" shall have the meaning attributed to it in section 5.25(c).
- 2.48 "Roof Deck" shall mean the layer of material (usually plywood or other similar material) installed on the roof trusses.
 - 2.49 "Settlement Class Members" shall mean members of the Settlement Classes.
- 2.50 "Settlement Classes" shall mean the Canadian National Class, the Quebec Class and the U.S. Class.
 - 2.51 "Settlement Notice" shall have the meaning attributed to it in section 9.1.
 - 2.52 "Settlement Website" shall have the meaning attributed to it in section 9.7.
- 2.53 "Settling Parties" shall mean the Defendant, the Plaintiffs and the Settlement Class Members.
 - 2.54 "U.S. Class" shall have the meaning attributed to it in section 3.3.
- 2.55 "U.S. Class Representatives" shall mean those class representatives known as Robert Melillo and Arthur Mayo Jr., who are collectively, the named Plaintiffs in the U.S. Action, individually and on behalf of members of the U.S. Class.
 - 2.56 "U.S. Court" shall mean the Vermont Superior Court, Chittenden Unit.
- 2.57 "U.S. Action" shall mean the putative class action lawsuit pending in the State of Vermont under the caption, Robert S. Melillo et al. v. Building Products of Canada Corp., Vermont Superior Court, Chittenden Unit, Court File No. s618-11 cnc.

3. <u>CLASS CERTIFICATION</u>

3.1 For the purpose of the Ontario Action, the Settling Parties agree that certification for settlement purposes of a settlement class defined as follows is appropriate (the "Canadian National Class"):

All individuals and entities, that own or owned, homes, residences, buildings, or other structures located in Canada whose roofs contain or contained BP Organic Shingles, excluding members of the Quebec Class.

3.2 For the purpose of the Quebec Action, the Settling Parties agree that authorization for settlement purposes of a settlement class defined as follows is appropriate (the "Quebec Class"):

All natural persons, as well as all legal persons established for a private interest, partnerships and associations having no more than 50 persons bound to it by contract of employment under its direction or control during the 12-month period preceding September 28, 2011, that own or owned, homes, residences, buildings, or other structures located in the Province of Quebec whose roofs contain or contained BP Organic Shingles.

3.3 For the purpose of the U.S. Action, the Settling Parties agree that certification of a settlement class for settlement purposes defined as follows is appropriate (the "U.S. Class"):

All individuals and entities, that own or owned, homes, residences, buildings, or other structures located in the United States whose roofs contain or contained BP Organic Shingles.

- 3.4 Excluded from the Settlement Classes are:
 - (a) all persons who timely exercise their rights under section 9 of the Class Proceedings Act, 1992, S.O. 1992 c. 6, article 1007 of the Quebec Code of Civil Procedure or V.R.C.P. 23 to opt out of the Settlement Class pursuant to the terms of this Agreement;
 - (b) all persons who filed a claim concerning their BP Organic Shingles in any court of law and whose claim has been resolved with a final judgment or settlement, whether or not favorable to the person; and

- (c) the Defendant, any entity in which the Defendant has a controlling interest, any entity that has a controlling interest in the Defendant, and the Defendant's legal representatives, assigns, and successors.
- 3.5 In the event this Agreement is terminated in accordance with its terms or otherwise fails to take effect for any reason, the Parties agree that any prior certification or authorization of an Action as a class proceeding pursuant to this Agreement, or any amended certification or authorization of an Action as a class proceeding pursuant to this Agreement, including the definition of the Settlement Class and the statement of any common issue, shall be without prejudice to any position that any of the Parties may later take on any issue in the Actions or any other litigation.

4. CLAIMS TO BE COMPENSATED BY THE AGREEMENT

Eligible Claims

- 4.1 Subject to section 4.2, a Claimant shall be deemed an "Eligible Claimant" and entitled to settlement benefits under this Agreement if:
 - (a) the Claimant is a member of the Settlement Class;
 - (b) if the Claimant is a former owner of a home, residence, building or other structure on which BP Organic Shingles are or were installed, the Claimant shall only be an Eligible Claimant if the Claimant: (i) retained from the subsequent property owner the rights and obligations created by this Agreement, in accordance with section 5.35(b) below; and (ii) otherwise satisfies all terms of this Agreement;
 - (c) if the Claimant is the current owner of a home, residence, building or other structure on which BP Organic Shingles are or were installed, the Claimant shall only be an Eligible Claimant if: (i) the Claimant has not assigned to the former property owner the rights and obligations created by this Agreement, in accordance with section 5.35(b) below; and (ii) the Claimant otherwise satisfies all terms of this Agreement;
 - (d) The Claimant files a duly completed and timely Claim Form in accordance with this Agreement, and prior to the removal of the BP Organic Shingles that are the subject of the claim from the home, residence, building or other structure on which they were installed;

- (e) Damage to BP Organic Shingles has occurred or occurs to the BP Organic Shingles applied to the Claimant's home, residence, building or other structure prior to the expiration of the Applicable Warranty Period; and
- (f) None of the Causation Defenses is established to be the principle cause of the Damage to BP Organic Shingles.
- 4.2 Any Claimant who has previously resolved a claim against the Defendant in relation to Damaged BP Organic Shingles, whether by settlement, adjudication or resolution of a warranty claim (including as evidenced by a written release of the Defendant, proof of compensation by the Defendant in settlement of the claim or an adjudication decision), relating to Damaged BP Organic Shingles shall not be compensated with respect to the same Damaged BP Organic Shingles, but can claim and be compensated with respect to Damaged BP Organic Shingles that were not the subject of the earlier settlement, adjudication or resolution of a warranty claim. Any Claimant who previously filed a warranty claim with the Defendant, and which warranty claim was denied before June 9, 2009, shall not be considered an Eligible Claimant, unless the conditions of the BP Organic Shingles that were the subject of the warranty claim have worsened, such that the conditions now qualify as Damage to BP Organic Shingles in accordance with section 4.6. This provision does not apply with respect to a Claimant that is claiming with respect to BP Organic Shingles that were not the subject of the previously-filed warranty claim.
- 4.3 Where a Claimant previously filed a warranty claim with the Defendant and which warranty claim was denied, in whole or in part, after June 9, 2009, and the Claimant has replaced the BP Organic Shingles that were the subject of the denied warranty claim, notwithstanding the requirement in section 5.4 that the Claim Form must be filed before the removal of the BP Organic Shingles, the Claimant shall be considered an Eligible Claimant.

Such claims shall be evaluated based on the information provided by the Claimant pursuant to the terms of this Settlement Agreement and, if reasonably available to the Defendant, the contents of the related warranty claim file, provided that the Defendant has sufficient information and documents to evaluate the claim. The claim may be denied where there is not sufficient information or documents for the Defendant to evaluate the claim.

- 4.4 Any Claimant who has previously received compensation as a result of an insurance claim or claim against a builder or other third party relating to Damaged BP Organic Shingles can claim under the Agreement for the same Damaged BP Organic Shingles, but any compensation accruable under the terms of the Agreement shall be offset by any amount paid to the Claimant as a result of such claim.
- 4.5 In the event that the Defendant becomes aware that it has received multiple claims relating to the same BP Organic Shingles before it has paid compensation on any of the multiple claims, the Defendant shall determine which Claimant, if any, is to be compensated. However, once the Defendant has provided compensation with respect to particular Damaged BP Organic Shingles, the Defendant shall not be obligated to provide any further compensation for the same Damaged BP Organic Shingles.

Damage to BP Organic Shingles

- 4.6 The following conditions of BP Organic Shingles will be deemed to constitute "Damage to BP Organic Shingles" and BP Organic Shingles that display such conditions will qualify as "Damaged BP Organic Shingles":
 - (a) an open hollow bump, 3/4" or 19 mm or more in diameter, in the coating layer of the shingle resulting in the underlying asphalt being weathered (that is, oxidized and dirty), also known as "blistering";

- (b) corners and edges of shingle tabs that are curled downward toward the Roof Deck raising the portion of the tab just interior to the edges by more than ½" or 12.7 mm, also known as "clawing";
- (c) cracks in the top-coating of the shingle penetrating through the organic felt that present a source for leakage, also known as "cracking";
- (d) tab corners that are raised above the plane of the Roof Deck by at least 1/2" or 12.7 mm for BP Organic Shingles manufactured more than ten years before the submission of the Claim Form, or by at least 3/8" or 9.5 mm for BP Organic Shingles manufactured less than ten years before the submission of the Claim Form, also known as "curling";
- (e) tab corners that are raised above the plane of the Roof Deck by at least 3/4" or 19 mm after being placed in a freezer at 0° Celsius for 15 minutes, also known as "cold weather curling";
- (f) puckers of at least 1/4" or 6.35 mm that appear along the side and bottom edges of the tabs, also known as "fishmouthing"; and
- (g) a loss of top surface of the shingle resulting in an exposure of the substrate equal in size to a dime, also known as "delamination".

Causation Defenses

- 4.7 The Defendant will not be liable for any Damage to BP Organic Shingles, to the extent that one or more of the following Causation Defenses was the principal cause of the Damage to BP Organic Shingles:
 - (a) Damage to BP Organic Shingles caused by and/or resulting from lightning, hurricanes, tornadoes, hailstorms, earthquakes, Act of God, fortuitous events, fire, flood, explosion, earth movement, or other *force majeure* events;
 - (b) Damage to BP Organic Shingles caused by settlement, distortion, failure, cracking or movement of the Roof Deck, walls or foundation of the building, as well as flashings or metal work;
 - (c) Damage to BP Organic Shingles caused by improper drainage, erosion, ordinary wear and tear, work performed on or to the roof (including the installation of any equipment on the roof), traffic on the roof, impact of foreign objects, or ice backup or ice damming;

- (d) Damage to BP Organic Shingles caused by any defects or failure of the material used as a Roof Deck over which the BP Organic Shingles have been applied;
- (e) Damage to BP Organic Shingles caused by the failure to install in accordance with the Defendant's published application instructions or good roofing practices, including the failure to install the shingles using the Defendant's cements or equivalent cements that meet applicable Can/CGSB or ASTM standards;
- (f) Damage to BP Organic Shingles caused by attic or roof ventilation that is not in compliance with applicable building codes, or by installing BP Organic Shingles directly over Roof Decks insulated with rigid insulation board, if there does not exist at least 1" or 2.54 cm of free-flowing air space between the Roof Deck to which the BP Organic Shingles are applied and the rigid insulation board;
- (g) Damage to BP Organic Shingles caused by the failure, on low slope roofs (between 2:12 and 4:12 pitch), to use appropriate BP Organic Shingles or to install ice and water underlayment or two layers of felt;
- (h) Damage to BP Organic Shingles caused by a design failure of the home or other building or the failure to comply with any applicable local, provincial, state or national building code, including without limitation the National Building Code of Canada and the International Building Code,
- (i) Damage to BP Organic Shingles caused by exposure to or application of paints or any improper cleaning solutions, coatings, or harsh air born or liquid chemicals or modifications of any kind;
- (j) Damage to BP Organic Shingles caused by failure to perform required roof maintenance or by improper roof maintenance such as pressure washing, failure to remove vegetation, moss, algae, fungus, lichens, mold or mildew growth from the roof, or failure to remove tree branches that are in contact with the roof. However, Damage to BP Organic Shingles caused by "blue-green" algae will only constitute a Causation Defense to the extent that the Applicable Warranty does not warrant against "blue-green" algae; and
- (k) Damage to BP Organic Shingles caused by intentional, reckless, or negligent conduct of a party other than the Defendant.

4.8 If the Defendant asserts any of the Causation Defenses described in section 4.7, the Defendant shall bear the burden of establishing that such defense was the principle cause of the Damage to BP Organic Shingles.

5. <u>CLAIMS PROCESS</u>

Role of the Defendant in Administering Claims

- 5.1 The Defendant will be responsible for administering claims under the Agreement. Claims shall be administered and resolved in a neutral, rational, responsive, cost-effective, and timely manner.
- 5.2 The Defendants will ensure that the following guidelines are observed throughout the claims administration process:
 - (a) The Defendant will assign French-speaking representatives to correspond with French-speaking Settlement Class Members and review Claim Forms filed in French;
 - (b) The Defendant will ensure that a dedicated toll-free telephone facility is created and operative in Canada and the United States for the purposes of providing information to Class Members regarding the Agreement and the claims process;
 - (c) The Defendant will ensure that the dedicated toll-free telephone facility is staffed with representatives who are bilingual in English and French and available to speak with Class Members weekdays, not including statutory holidays, between 8:30am and 4:00pm EST;
 - (d) The Defendant will create an automated message with relevant information for Class Members and will allow Class Members to leave voice messages in the event that a representative is not available between the hours of 8:30am and 4:00pm EST to take the call. To the extent that calls are received outside business hours, the automated message shall advise Class Members to call during business hours in order to speak with a representative or leave a voice message; and
 - (e) The Defendant will direct its representatives to use reasonable best efforts to respond to any voice messages within two (2) business days.

Submission of Claim Forms

- 5.3 Claim Forms shall be submitted by Claimants to the Defendant at the address specified in the Claim Form. Claim Forms sent to an incorrect address will not be considered and will not constitute valid Claim Forms.
- 5.4 Except as otherwise provided for in this Agreement, Claim Forms must be postmarked no later than 150 days after the later of the Effective Date of this Agreement and the Claimant's discovery of the defect or need for any repairs or replacement and, in any event, no later than the expiry of the Applicable Warranty Period and prior to the removal of the BP Organic Shingles that are the subject of the claim from the home, residence, building or other structure on which they were installed.
- 5.5 Claim Forms can be filed in the English or French languages. Where a Claimant has filed a Claim Form in the French language, any subsequent correspondence with the Claimant shall be in the French language.
- 5.6 As part of the Claim Form, Claimants shall be required to provide, as to the property or properties to which the claim regarding BP Organic Shingles relates, all the following information and documents. The failure to provide all of the following information and documents may result in the denial of a claim:
 - (a) Name of Claimant and any co-owners.
 - (b) Address (including an email address) of Claimant and any co-owners.
 - (c) Address of property upon which the BP Organic Shingles that are the subject of the claim are installed (if different than the address of the Claimant and any co-owners).
 - (d) Photographs that establish the condition of the shingles in sufficient detail and quality such that evaluation of the claim can be made. All photographs should be labeled by the Claimant with the Claimant's name and identify the slope or area of the roof shown. The Claimant shall make

reasonable best efforts to provide each of the following: (i) one or more photographs showing the structure from a distance sufficient to show the entire structure including views of all roof vents; (ii) one or more photographs showing the inside and outside of all of the soffits; and (iii) a minimum of two photographs of each roof slope showing the slope and the condition of the shingles. The photographs provided must be of sufficient detail and clarity to enable the Defendant to evaluate the claim. Photographs can be submitted either in hard copy or electronically.

- (e) Reliable and contemporaneous documentary proof of purchase and installation of BP Organic Shingles, such as an invoice from a roofer for services already rendered (bids, quotations or estimates shall not be acceptable), or if the BP Organic Shingles are the original installation on the structure, a certificate of occupancy, a final building inspection or an invoice from the builder or roofing contractor stating the date of installation. If the documentary proof of purchase and installation establishes the year but not the month of installation, the installation date shall default to the earliest month of the year.
- (f) A good faith estimate of the number of Damaged BP Organic Shingles.
- (g) The claimant shall make reasonable best efforts to provide information regarding the measurement of the roof including the following:
 - (i) detailed measurements of the roof, on a slope-by-slope basis;
 - (ii) the number of roof bundles or squares installed on the roof. Where the number of roof bundles or squares installed on the roof is not available, the square footage of the relevant roof structure, along with a measurement of the square footage of the ground floor of the structure, the number of stories, and the pitch or pitches of the roof; and
 - (iii) all other relevant information in the Claimant's possession that might assist the Defendant in its determination of the measurements of the roof.
- (h) A statement that the Claimant authorizes an inspection of the structure(s) by the Defendant and/or the Inspector in accordance with the terms of this Agreement.
- (i) Whether the property was the subject of an insurance claim or claim to a builder or other party relating to any Damage to BP Organic Shingles and, if so, evidence of when the claim was made, against whom the claim was made and the amount of any compensation received in respect of the claim.

- (j) Whether there has been an assignment of any claims relating to BP Organic Shingles and documentation of such assignment.
- (k) Whether the Claimant is electing the Payment of Repairs Settlement Option or the Cash Settlement Option.
- (l) A statement that the Claimant agrees to cooperate with the Defendant in providing any reasonably requested additional information required to determine the validity of the claim.
- (m) A declaration by the Claimant and any co-owners, under penalty of perjury, that information or material submitted to the Defendant is true and correct.
- (n) The signature of the Claimant and any co-owners.
- 5.7 As part of the Claim Form, Claimants may also be required to provide, as to the property or properties to which the claim regarding BP Organic Shingles relates, substantially all the following information and documents:
 - (a) Date that each structure was built, if known.
 - (b) Builder (if original installation) or roofing contractor (if remodel or replacement), if known.
 - (c) Development, neighbourhood, or subdivision, if known.
 - (d) Date of purchase of the property.
 - (e) Two forms of verification of current or former ownership of the property. Such verification includes a copy of a property deed, property tax record, mortgage statement, property insurance bill, or utility bill. For manufactured homes, such verification can also include a copy of the registration or title.
 - (f) Type of structure(s) upon which the BP Organic Shingles that are the subject of the claim are installed.
 - (g) A prior communication from the Defendant confirming that the shingles on the structure are BP Organic Shingles, if such communication exists.
 - (h) A report from a reputable roofing contractor documenting the condition of the BP Organic Shingles, including any Damage to BP Organic Shingles, will be required. The report must contain sufficient detail such that an evaluation of the claim can be made.

- (i) Whether the property has any features that might interfere with an inspection, such as locked gates or dogs.
- (j) Whether the Claimant or a prior owner previously made a claim to the Defendant with respect to the BP Organic Shingles, the approximate date of such claim, and the outcome of the claim, including the amount of any compensation received.
- (k) Whether the Claimant has previously filed a Claim Form.

Where a Claimant is unable to provide one or more of the items listed above, the Claimant shall be permitted to furnish alternative proof in order to support the claim, provided that the alternative proof is sufficient to enable the Defendant to evaluate the claim. The failure to provide sufficient information and documents to enable the Defendant to evaluate the claim may result in the denial of a claim.

5.8 BP can request that the Claimant provide any other information or documents that might be necessary to assist BP in the evaluation of claims. BP can also request that the Claimant provide a sample of at least one full size shingle certified by the Claimant to be from the structure and of sufficient condition such than an evaluation of the claim can be made. The Claimant must also provide photographs indicating the location from which the sample was removed. Photographs must show the location both before and after the removal of the shingle, and must be labeled with the Claimant's name and identify the slope or area of the roof shown. Photographs can be submitted either in hard copy or electronically. Where the Claimant provides invoices establishing expenses incurred in having a roofing professional remove and replace the shingles necessary to provide the sample shingle to the Defendant, including shipping, the Defendant will reimburse Claimants who are members of the Canadian National Class or Quebec Class up to CDN \$50.00 and Claimants who are members of the U.S. Class up to U.S. \$50.00 for such expenses, including shipping costs, provided that the Claimant is found

to be a Settlement Class Member and the shingles, upon inspection and testing, are determined to be BP Organic Shingles. Such reimbursement shall be paid by cheque and shall be included with the Claim Decision.

- 5.9 The failure of the Claimant to cooperate with the Defendant in a reasonable manner in providing sufficient information and documents for the Defendants to evaluate the claim, including the provision of sample shingles in accordance with section 5.8, shall result in the denial of the claim.
- 5.10 No materials submitted by any Claimant, including shingle samples and photographs, will be returned to the Claimant.
- 5.11 Claimants shall provide the Defendant with any change of address or updated or revised contact information.
- 5.12 Claimants cannot utilize third-party claims services or similar services to file claims. Where Claimants utilize third-party claims services or similar services to file claims, their claims will be treated as deficient and they will be given an opportunity to cure the deficiency in accordance with section 5.16 below. Notwithstanding the foregoing, nothing in this section shall preclude Settlement Class Members from being assisted by Class Counsel or personal counsel of their own choosing in the completion of their Claim Form and pursuit of their claims.
- 5.13 Where a claim has been denied, the Claimant or any eligible successor may not submit another claim with respect to the BP Organic Shingles that were the subject of the claim for two years from the date of the Claim Decision, unless the Claimant or any eligible successor is experiencing actual water leakage as a result of a worsening of the conditions of the BP Organic Shingles, in which case the Claimant or any eligible successor can file a subsequent

claim immediately upon discovering the leak. Thereafter, the Claimant or eligible successor may file a subsequent claim provided that the subsequent claim is postmarked on or before the expiration of the Applicable Warranty Period. If a subsequent claim is accepted, the settlement benefits accruable to the Claimant shall be calculated based upon the date of the submission of the subsequent claim.

Processing of Claim Forms

- 5.14 The Defendant shall scan each Claim Form and any supporting documentation into a database established for this Agreement or shall otherwise manage the information in an efficient and accessible manner. The information shall be available to Class Counsel in such form as Class Counsel and the Defendant shall agree.
- 5.15 Any communications required in the administration of a claim may be sent by email where the Claimant has provided his/her email address.
- 5.16 The Defendant shall review the Claim Form and any supporting documentation to determine whether the Claimant is a Settlement Class Member and whether the Claim Form is complete and includes all the required supporting documentation necessary to proceed with the processing of the claim. If the Claimant is not a Settlement Class Member, the Defendant shall send a letter to the Claimant notifying the Claimant of that fact. If the Claimant is a Settlement Class Member, but the Claim Form and supporting documentation is deficient, the Defendant shall send a deficiency letter to the Claimant identifying the deficiencies and providing the Claimant 60 days to cure the deficiencies. The letter shall advise the Claimant that if the Claimant does not resolve the identified deficiencies within 60 days from the date of the letter, the claim shall be denied. If the Claimant does not resolve the identified deficiencies within 60 days from the date of the letter, the claim shall be denied, without prejudice to the Claimant's

right to file another Claim Form in accordance with the terms of this Agreement, and the Defendant shall promptly issue a Claim Decision denying the claim. The Claim Decision shall notify the Claimant of the denial of the claim, the Claimant's right to file another Claim Form in accordance with the terms of this Agreement and the Claimant's right to seek a review of the Claim Decision in accordance with the terms of this Agreement.

- 5.17 Once a completed and timely Claim Form has been received by the Defendant, it shall evaluate the claim and determine whether the Claimant is an Eligible Claimant. The Defendant may contact the Claimant in connection with its processing and evaluation of the claim.
- 5.18 The Parties anticipate that claims may be evaluated based on the information provided by the Claimant pursuant to sections 5.6, 5.7 and 5.8. However, the Defendant may inspect any BP Organic Shingles that are the subject of a claim if, in the Defendant's determination, such inspection is reasonably necessary. The Defendant shall make reasonable efforts to notify the Claimant of the date and approximate time of the inspection. The Claimant can be present during the inspection, as scheduled by the Defendant. In conducting any inspection, the Defendant shall be permitted interior access to the roof and attic area of the relevant structure. The failure of the Claimant to provide reasonable interior access to the roof and attic area of the relevant structure shall result in the denial of the claim. The Defendant shall have the right to remove such shingles, soffits, vents, fascias, rake boards, or other portions of the structure as may be reasonably necessary to determine whether there exists a Causation Defense, provided, however, that following any such removal the Defendant shall, at its own expense, restore the Claimant's property to the condition that existed prior to the inspection. In the event an inspection is needed, the Defendant shall use reasonable best efforts to schedule

such inspection within 60 days of the receipt of a completed Claim Form, but shall be granted additional time in the event that weather conditions or volume of claims affect the Defendant's ability to timely proceed. Promptly after conducting the inspection, the Defendant shall prepare an inspection report. The Defendant shall indemnify and hold harmless Claimants for any injuries to the Defendant during the course of any inspection pursuant to this section, except where such injuries arise from willful conduct or negligence of the Claimants or other persons on the Claimants' property or otherwise under the Claimants' direction or control.

- 5.19 Within 120 days of receiving the later of (i) a completed Claim Form, (ii) any additional information requested pursuant to sections 5.7 and 5.8 above, or (iii) an inspection report prepared pursuant to section 5.18, the Defendant shall make a Claim Decision.
- 5.20 The Defendant shall send a copy of the Claim Decision to the Claimant promptly after making the Claim Decision. In addition, where a claim has been rejected in whole or in part, the Defendant shall promptly send a copy of the Claim Decision by email to Charles Laduca or such other individual as identified by Class Counsel.
- 5.21 Where a claim has been rejected in whole or in part, the Claim Decision shall include a brief written explanation of the basis of the Claim Decision and information explaining the right of a Claimant to seek a review of the Claim Decision, in accordance with the terms of this Agreement.
- 5.22 Where a claim has been accepted in whole or in part and the Claimant has elected the Payment of Repairs Settlement Option, the Defendant shall include in the Claim Decision the information required in section 6.4(a) below.

- 5.23 Where a claim has been accepted in whole or in part and the Claimant has elected the Cash Settlement Option, the Defendant shall include with the Claim Decision, the settlement benefits to which the Claimant is entitled pursuant to section 6.5 below.
- 5.24 The Defendant shall use reasonable diligence to locate Claimants whose Claim Decision is returned as undeliverable. Any such funds arising or remaining from any undeliverable cheques from any claim shall be the Defendant's property.

Review of Claims Decision

- 5.25 Where a claim has been denied, in whole or in part, or where the Claimant disputes the extent of Damaged BP Organic Shingles on a roof slope, the reasonableness of the Approved Repairs Amount or the proration of the Approved Repairs Amount, the Claimant has the right to request a review of the Claim Decision. A Claimant shall not have the right to request a review where such request for review relates solely to the interpretation or applicability of a term of this Agreement. The following procedures shall govern all reviews:
 - (a) Where the Claimant disputes the Claim Decision, the Claimant can seek a review of the Claims Decision by submitting a fully completed Request for Review Form to the Defendant, postmarked no later than 35 days from the date of the Claim Decision. The Claimant is not permitted to append or include any information or documents that were not submitted by the Claimant with the Claim Form or in response to any requests for additional information from the Defendant.
 - (b) Upon receipt of a Request for Review Form, the Defendant shall have the opportunity to re-consider the Claims Decision and to accept all or part of the claim in its sole discretion. If the Defendant accepts all or part of the claim, the Defendant shall issue a new Claim Decision in accordance with the terms of this Agreement.
 - (c) Within 45 days of receiving a Receipt for Review Form, if the Defendant does not accept all or part of the claim, the Defendant shall prepare and provide the Inspector or the Adjudicator, as appropriate, with the following "Review Materials": (i) the Claim Form and any supporting documents; (ii) any information or documents that were submitted by the

Claimant in response to requests for additional information from the Defendant; (iii) any inspection report prepared by the Defendant or an agent of the Defendant; (iv) the Claims Decision; (v) the Request for Review Form; and (vi) a response to Request for Review setting out the Defendant's response to any statements contained in the Request for Review Form.

- (d) If the Inspector, in its sole discretion, deems it necessary to inspect a roof, the Inspector can visit the property to inspect the roof. The Inspector shall make reasonable efforts to notify the Claimant of the date and approximate time of the inspection. The Claimant can be present during the inspection, as scheduled by the Inspector. The Inspector shall conduct the inspection within 45 days of receiving the Review Materials, but shall be granted additional time in the event that weather conditions or volume of Requests for Review affect the Inspector's ability to timely proceed. In conducting any inspection, the Inspector shall be permitted interior access to the roof and attic area of the relevant structure. The failure of the Claimant to provide the Inspector reasonable interior access to the roof and attic area of the relevant structure, when requested by the Inspector, shall result in the denial of the claim.
- (e) Where the review is conducted by the Adjudicator, the Adjudicator shall conduct a review of the Claim Decision as expeditiously as possible, and in any event within 45 days of receipt of the Review Materials, to make a Review Decision. The Adjudicator's Review Decision shall be based solely on the terms of this Agreement and the Review Materials.
- (f) Where the review is conducted by the Inspector, the Inspector shall conduct review of the Claim Decision as expeditiously as possible, and in any event within 45 days of receipt of the Review Materials, or if an inspection was conducted, within 45 days of conducting the inspection, to make a Review Decision. The Inspector's Review Decision shall be based solely on the terms of this Agreement, the Review Materials, and the results of any inspection conducted by the Inspector.
- (g) For greater certainty, the Inspector and the Adjudicator have no authority to award to a Claimant any relief other than the settlement benefits provided for in this Agreement.
- (h) Promptly after making a Review Decision, the Inspector or the Adjudicator shall send copies of the Review Decision to the Claimant and the Defendant.
- (i) Where the Claimant has elected the Cash Settlement Option, within 30 days of receipt of the Review Decision, the Defendant shall provide to the

- Claimant any settlement benefits due and owing as a result of the Review Decision.
- (j) Where the Claimant has elected the Payment of Repairs Settlement Option, within 120 days after the receipt of the Review Decision, the Defendant shall provide to the Claimant the information required pursuant to section 6.4(a) below.
- (k) The Review Decision shall be final and binding and not subject to any further right of appeal.
- 5.26 Where a Claimant has filed a Claim Form in the French language, the Review Decision shall be issued in the French language.
- 5.27 The Inspector may review disputed Claims Decisions, where the dispute relates to: (i) whether the shingles at issue are BP Organic Shingles; (ii) whether there is Damage to BP Organic Shingles; (iii) whether the Defendant has established that a Causation Defense was a principal cause of the Damage to BP Organic Shingles; (iv) measurement of the roof slope; (v) the extent of Damaged BP Organic Shingles on a roof slope; (vi) the reasonableness of the Approved Repairs Amount; and/or (vii) the proration of the Approved Repairs Amount. The Inspector's jurisdiction in reviewing disputed Claims is limited by section 5.25(g).
- 5.28 The Adjudicator may review any disputed Claim Decisions, where the dispute relates to whether the Claimant has filed a properly completed Claim Form or any matter that is not enumerated in section 5.27. The Adjudicator's jurisdiction in reviewing disputed Claims is limited by section 5.25(g).
- 5.29 The Adjudicator or Inspector, as the case may be, shall deny a claim where the shingles are not BP Organic Shingles or where the Claimant has engaged in fraudulent and/or abusive practices, including but not limited to submitting false claims, samples or documentation, and shall take such other actions as may be appropriate to prevent such

fraudulent and/or abusive practices in the future. Class Counsel and the Defendant shall cooperate in discouraging any abuse of the claims process.

- 5.30 Where the Adjudicator or Inspector, as the case may be, determines that a Claimant has engaged in fraudulent practices or that a Claimant requested a review of the Claim Decision where the shingles are not BP Organic Shingles, the Adjudicator or Inspector shall award costs payable forthwith to the Defendant by the Claimant, in the amount of CDN \$250.00 for Claimants who are members of the Canadian National Class or the Quebec Class, and U.S. \$250.00 for Claimants who are members of the U.S. Class. There shall be no right of appeal from a decision awarding costs to the Defendant pursuant to this section.
- 5.31 The Inspector and the Adjudicator shall maintain records of their activities in a computerized database electronically accessible by Class Counsel and the Defendant in a secure, read-only environment and shall provide such periodic and special reports as the Courts, Class Counsel, or the Defendant may request.
- 5.32 The Adjudicator and the Inspector shall each have a continuing obligation to be neutral and unbiased in respect of their obligations under this Agreement for the duration of their respective appointments. Defendant and Class Counsel shall have the right to audit independently the work of the Inspector and the Adjudicator. If the Defendant and/or Class Counsel are not satisfied that the Inspector and/or the Adjudicator are resolving Requests for Review in a neutral, fair and prompt manner, they shall work with the Inspector and/or the Adjudicator to resolve any issues. If the issue(s) is not resolved in an appropriate manner, the Defendant and/or Class Counsel can ask for the appointment of a new entity, to be agreed upon by the Parties, to perform the duties of the Inspector and/or the Adjudicator under this Agreement. If the Defendant and Class Counsel are unable to reach agreement on whether it is

necessary to replace the Inspector and/or the Adjudicator or who should be appointed as a replacement for the Inspector and/or the Adjudicator, the Party seeking to replace the Inspector and/or the Adjudicator shall seek directions from the Ontario Court in this regard. The Ontario Court's decision in this respect shall not be subject to any right of appeal.

5.33 In no event shall any Released Party have any liability for claims of wrongful or negligent conduct on the part of the Inspector, the Adjudicator or their respective agents.

Assignment of Claims and Disclosure to Subsequent Property Owners

- 5.34 Unless the relevant Damaged BP Organic Shingles have been removed and replaced, any Claimant who has received settlement benefits pursuant to the Cash Settlement Option, in respect of a claim, shall advise any subsequent purchaser of the property in respect of which the claim was made of: (i) the fact that settlement benefits have been received; (ii) that no additional claim can be made under the Agreement with respect to those Damaged BP Organic Shingles for which settlement benefits have been received; and (iii) such other disclosures as may be required by law.
- 5.35 Except as provided in this section, Settlement Class Members may not assign their claims.
 - (a) Subject to subsection (b) and to the terms of the applicable warranty, upon the sale of a property with a building with BP Organic Shingles, the purchaser shall succeed to the rights of the Settlement Class Member by acquiring such property and will receive and succeed to all rights and obligations created by this Agreement.
 - (b) Upon the sale of a property with a building with BP Organic Shingles, a Settlement Class Member who sells the property may retain, pursuant to a written assignment agreement executed contemporaneously with the sale of the property, all rights and obligations created by this Agreement. In these circumstances, the Settlement Class Member must submit a Claim Form postmarked no later than 60 days after the later of the Effective Date

or the sale of the property and include evidence of the assignment with the Claim Form.

Costs of Administering Claims

5.36 Except as otherwise provided by this Agreement, the Defendant shall be solely responsible for and shall pay all reasonable fees and expenses incurred by the Inspector and the Adjudicator in administering Requests for Review and disputes relating to Repair Invoices pursuant to section 6.4(h) as well as all costs of the Defendant incurred in administering claims.

Reporting and Audit Rights

- 5.37 On the first anniversary of the Effective Date, and annually thereafter until one year after the expiration of the last of the Applicable Warranty Periods, the Defendant shall serve on a designee of Class Counsel a report identifying the Claimants whose claims have been resolved in the prior 12 months and the settlement benefits distributed to each Eligible Claimant.
- 5.38 Class Counsel shall have the right to audit, on an annual basis, the processing and disposition of claims by the Defendant. In connection with such an audit, Class Counsel shall have the right to examine all books and records maintained by the Defendant related to the processing of claims covered by this Agreement, including any Claim Forms, inspection reports and correspondence with Claimants.
- 5.39 The Defendant shall maintain a copy of all paper and electronic documents or records maintained by the Defendant in respect of claims submitted until two years after the relevant claim has been finally resolved. However, where a claim has been allowed in full, the Defendant can immediately dispose of any shingle samples submitted by the Claimant. Where a claim has been denied, in whole or in part, the Defendant must retain a 1 foot square exemplar of any shingle samples submitted by the Claimant until 45 days after the issuance of a Claim

Decision, or, if the Claimant has filed a timely Request for Review, until 45 days after the disposition of the independent review.

5.40 Class Counsel and counsel for the Defendant shall meet in person or by telephone conference, as reasonably necessary, to discuss the implementation of this Agreement and attempt to resolve any concerns of the Parties. If the Parties are unable to resolve their concerns in accordance with this section, such disputes shall be resolved in accordance with section 16.6.

Confidentiality of Claimant Information

5.41 Information and documents about individual claims, claims processing and inspections is confidential and shall only be disclosed in accordance with the terms of this Agreement, any order of the Courts, or as otherwise required by the law.

6. <u>SETTLEMENT BENEFITS</u>

- 6.1 If the Defendant, the Inspector or the Adjudicator determines that the Claimant is an Eligible Claimant, settlement benefits shall be issued pursuant to sections 6.2 to 6.6 below.
- 6.2 The number of squares on which settlement benefits shall be calculated is based on the size of the roof slope and the extent of Damaged BP Organic Shingles on a roof slope, as follows:
 - (a) If the Damaged BP Organic Shingles comprise five (5%) percent or more of a roof slope, settlement benefits will be calculated based on the total number of squares on the entire roof slope.
 - (b) If the Damaged BP Organic Shingles comprise less than five (5%) percent of a roof slope, settlement benefits will be calculated based on the number of Damaged BP Organic Shingles.
 - (c) If an Eligible Claimant receives settlement benefits for less than the entire roof slope and the Claimant suffers additional Damage to BP Organic Shingles to other portions of the roof slope, the Claimant can submit further claims with respect to such other portions of the roof slope in accordance with the terms of this Agreement.

While the number of squares forms the basis on which settlement benefits are calculated, settlement benefits will be prorated, if applicable, in accordance with sections 6.4(b), 6.5 and 6.6 below.

- 6.3 Eligible Claimants shall have two settlement benefit options: the Payment of Repairs Settlement Option and the Cash Settlement Option.
- 6.4 Under the Payment of Repairs Settlement Option, subject to section 4.3, the Defendant shall provide settlement benefits to Eligible Claimants, in accordance with the following provisions:
 - (a) The Claim Decision or Review Decision shall state the extent to which repair or replacement of the Damaged BP Organic Shingles will be compensated pursuant to the terms of this Agreement. In particular, the Claim Decision or Review Decision shall state the number of squares (calculated in accordance with section 6.2) with respect to which the Defendant shall be required to pay all or part of the costs of repairing or replacing the Damaged BP Organic Shingles (the "Approved Repairs"). The Claim Decision or notice provided in accordance with section 5.25(j) shall also state:
 - (i) the Defendant's estimate of the costs of repair or replacement in Canadian (CDN) dollars for members of the Canadian National Class or the Quebec Class and American (U.S.) dollars for members of the U.S. Class. Such estimate shall be calculated based on the following principles:
 - (1) Subject to the other provisions of this section 6.4(1)(a)(i), the estimate shall be calculated in accordance with the terms of the Applicable Warranty and, for greater clarity, shall not include costs in respect of repairs or replacement that are not specifically provided for under the terms of the Applicable Warranty.
 - (2) During the proration period only, such estimate shall be based on the costs of repairing or replacing the Damaged BP Organic Shingles at the time that the claim was filed in accordance with the terms of this Agreement and not at the time of the original installation of the Damaged BP Organic Shingles.

(3) In accordance with the terms of the Applicable Warranty, notwithstanding the foregoing, for BP Organic Shingles installed on or after January 1, 2006, such estimate shall be no greater than the original costs of installing the Damaged BP Organic Shingles.

The estimate calculated in accordance with this section 6.4(a)(i) shall represent the total possible settlement benefits payable under the Payment of Repairs Settlement Option (the "Approved Repairs Amount");

- (ii) the portion of the Approved Repairs Amount that will be reimbursed by the Defendants, provided that the Claimant otherwise satisfies the requirements of this Agreement;
- (iii) instructions on submitting Repairs Invoices for reimbursement; and
- (iv) instructions on how to request a review with respect to the reasonableness of the Approved Repairs Amount and/or the proration of the Approved Repairs Amount. Any disputes regarding the reasonableness of the Approved Repairs Amount or the proration of the Approved Repairs Amount shall be resolved by the Inspector in accordance with section 5.25 above.
- (b) During the Non-Proration Period, the Defendant shall reimburse the Claimant for the full Approved Repairs Amount. After the Non-Proration Period, the Defendant shall reimburse the Claimant for the Approved Repairs Amount on a prorated basis, calculated in accordance with section 6.6 below. Except as provided for in this subsection, the Defendant shall have no responsibilities in respect to costs incurred by the Claimant in repairing and/or replacing the Damaged BP Organic Shingles.
- (c) The Claimant must complete the Approved Repairs within one year of the date of the Claim Decision or notice provided in accordance with section 5.25(j). In completing the Approved Repairs, the Eligible Claimant can select the roofer of his/her/its choice. However, the Eligible Claimant must use roofing shingles manufactured by the Defendant. The failure by the Eligible Claimant to use roofing shingles manufactured by the Defendant will result in a forfeiture of settlement benefits in respect of the BP Organic Shingles that are the subject of the claim and will relieve the Defendant of its obligation to pay any portion of Approved Repairs Amount.
- (d) The Eligible Claimant must submit to the Defendant copies of Repairs Invoices immediately upon receipt of such Repairs Invoices and in any

event no later than 15 days after receipt of such Repairs Invoices. The Repairs Invoices must clearly indicate the type of shingles installed on the structure. The failure by the Eligible Claimant to provide adequate documentation establishing that the Eligible Claimant used roofing shingles manufactured by the Defendant will result in a forfeiture of settlement benefits in respect of the BP Organic Shingles that are the subject of the claim and will relieve the Defendant of its obligation to pay any portion of Approved Repairs Amount.

- (e) Within 30 days of receipt of Repairs Invoices, where the Defendant is satisfied that all or part of the Repairs Invoices relates to Approved Repairs and the Eligible Claimant used roofing shingles manufactured by the Defendant, the Defendant shall reimburse the Eligible Claimant for the portion of the Approved Repairs Amount that is payable by the Defendant pursuant to sections 6.4(b) and 6.6. Payments shall be issued in Canadian (CDN) dollars for members of the Canadian National Class or the Quebec Class and in American (U.S.) dollars for members of the U.S. Class.
- (f) Within 30 days of receipt of Repairs Invoice, where the Defendant takes the position that all or part of a Repairs Invoice does not relate to Approved Repairs or the Repairs Invoices do not adequately establish that the Eligible Claimant used roofing shingles manufactured by the Defendant, the Defendant shall notify the Claimant of the following:
 - (i) The Defendant's position that all or part of the Repairs Invoice does not relate to Approved Repairs or the Repairs Invoice does not adequately establish that the Eligible Claimant used roofing shingles manufactured by the Defendant; and
 - (ii) the Claimant has the right to seek review of the Defendant's position by submitting a Repair Dispute Form to the Defendant, postmarked no later than 15 days of the notice.
- (g) Upon receipt of a Repair Dispute Form, the Defendant shall have the opportunity to re-consider the Repairs Invoices and to reimburse the Eligible Claimant for all or part of the Repairs Invoice in its sole discretion. If the Defendant only accepts part of the Repairs Invoice, the Defendant shall issue a notice to the Eligible Claimant in accordance with section 6.4(f).
- (h) Within 15 days of receipt of a Repair Dispute Form, if the Defendant does not reimburse the Eligible Claimant for all or part of the Repairs Invoice in accordance with section 6.4(g), the Defendant shall forward the Repair Dispute Form to the Inspector, along with a copy of the Claim Decision or Review Decision, the notice referred to in section 6.4(f) and the Repairs Invoice. The Inspector shall promptly and, in any event within 30 days of

receipt of such material, resolve such disputes based solely on a review of the Claim Decision or Review Decision, the notice referred to in section 6.4(f) and the Repairs Invoice and the terms of this Agreement. The Inspector shall promptly notify the Defendant and the Claimant of its decision regarding the eligibility of the repairs. Where the Inspector determines that the Repairs Invoice relates to Approved Repairs or that the Repairs Invoice adequately establishes that the Eligible Claimant used roofing shingles manufactured by the Defendant, the Defendant shall within 45 days of receipt of the Inspector's decision reimburse the Claimant in respect of the Repairs Invoice in accordance with the Inspector's decision and the terms of this Agreement. The Inspector's decision shall be final and binding and subject to no further right of appeal.

- 6.5 Under the Cash Settlement Option, subject to section 4.3, the Defendant shall provide settlement benefits to Eligible Claimants in accordance with the following provisions:
 - (a) The Claim Decision or Review Decision shall state the number of squares (calculated in accordance with section 6.2) with respect to which the Defendant shall be required to pay;
 - (b) During the Non-Proration Period, the Defendant shall pay Eligible Claimants \$75/square, which amount constitutes compensation for the costs of replacement shingles and any labour expenses associated with removing and replacing the Damaged BP Organic Shingles. Payments shall be issued in Canadian (CDN) dollars for members of the Canadian National Class or the Quebec Class, and in American (U.S.) dollars for members of the U.S. Class; and
 - (c) After the Non-Proration Period, the Defendant shall pay Eligible Claimants \$75/square on a prorated basis, calculated in accordance with section 6.6 below, which amount constitutes compensation for the costs of replacement shingles and any labour expenses associated with removing and replacing the Damaged BP Organic Shingles. Payments shall be issued in Canadian (CDN) dollars for members of the Canadian National Class or the Quebec Class, and in American (U.S.) dollars for members of the U.S. Class.
- 6.6 After the Non-Proration Period, all compensation to be calculated pursuant to sections 6.4(b) and 6.5 will be prorated based on the proration percentages contained in the Applicable Warranty. The proration shall be calculated based on the date that a completed Claim Form is received by the Defendant, except in circumstances where the Claimant has previously

filed a warranty claim with the Defendant, which previously filed warranty claim was rejected in whole or in part, and it is determined that, had the previously filed warranty claim been determined in accordance with the terms of this Settlement Agreement, the Claimant would have been entitled to compensation, in which circumstances the proration shall be calculated based on the date that the previously-filed warranty claim was received, in full, by the Defendant.

7. <u>CLASS COUNSEL FEES AND EXPENSES</u>

7.1 The Defendant shall pay the reasonable legal fees, disbursements and applicable Canadian provincial and federal sales taxes of all counsel for the Plaintiffs in the Actions in the amount of Cdn \$2,400,000, subject to the approval of the Courts. The Defendant shall have no further obligation in respect of the present or future legal fees, disbursements, or taxes of counsel for the Plaintiffs in the Actions.

8. PRELIMINARY APPROVAL ORDERS

- 8.1 Class Counsel shall file motions for "Preliminary Approval Orders" as soon as practical after the execution of this Agreement, requesting that the Courts enter Preliminary Approval Orders sequentially, first in the Ontario Court, and then in the Quebec Court and then in the U.S. Court, that:
 - (a) Provide for the conditional certification or authorization of the Settlement Classes, as applicable, for settlement purposes only, pursuant to the Class Proceedings Act, 1992, S.O. 1992, c. 6 in respect of the Canadian National Class, the Quebec Code of Civil Procedure in respect of the Quebec Class and V.R.C.P. 23 in respect of the U.S. Class, and approval of one or more of the Canadian Class Representatives to act as the representative plaintiffs for the Canadian National Class or the Quebec Class, as appropriate, and one or more of the U.S. Class Representatives to act as representative plaintiff(s) for the U.S. Class;
 - (b) Approve the form and content of the Settlement Notice and the method of disseminating the Settlement Notice;

- (c) Approve the process and set the deadline for Settlement Class Members to opt out of the applicable Settlement Class or object to the Agreement; and
- (d) Schedule dates for Final Approval Hearings.
- 8.2 Prior to and as a condition precedent to filing with the Courts, Class Counsel shall obtain the signed, written approval of counsel for the Defendant as to the form and content of the motion materials and orders contemplated by section 8.1.

9. NOTICE OF SETTLEMENT

- 9.1 The summary and long-form "Settlement Notices" shall be in a form to be agreed upon by the Parties or in such other form as approved by the Courts, and shall: (i) advise Settlement Class Members of the contingent certification or authorization of the Settlement Classes; (ii) state that the Agreement is contingent upon the Courts' final approval; (iii) advise Settlement Class Members of their rights to opt out of the Settlement Class, object to the Agreement, file a claim under the settlement, or do nothing, and the timelines, process and consequences of each; (iv) advise Settlement Class Members of the dates, times and locations of the Final Approval Hearings; and (v) provide instructions as how to obtain additional information about the Agreement and Settlement Class Members' options with respect to the Agreement. Prior to and as a condition precedent to seeking the approval of the Courts of the Settlement Notices, Class Counsel shall obtain the signed, written approval of Counsel for the Defendant as to the form and content of such Settlement Notices.
- 9.2 The summary Settlement Notice shall be published in such newspaper and other publications as agreed upon by the Parties or as otherwise ordered by the Courts.
- 9.3 The long-form Settlement Notice shall be mailed, first class postage prepaid, to Settlement Class Members identified by the Parties through reasonable efforts and reasonably

identifiable distributors of BP Organic Shingles, for the purposes of providing notice to their customers who might be Settlement Class Members, at the addresses last known to the Defendant.

- 9.4 The Defendant or its agents shall be responsible for the dissemination of the Settlement Notices, including all costs associated with disseminating the Settlement Notices (such as the costs of printing, mailing, and publication).
- 9.5 Within 30 days of dissemination of the Settlement Notices in accordance with this Agreement and any orders of the Courts, the Defendant shall provide to Class Counsel written confirmation that the Settlement Notices have been disseminated in accordance with this Agreement and any orders of the Courts.
- 9.6 No later than the publication of the first summary Settlement Notice pursuant to section 9.2, the Defendant shall cause a Canada and United States-wide toll-free telephone facility to be established, which may, in the sole discretion of the Defendant, be an existing toll-free telephone facility. The toll-free telephone number shall be included in the Settlement Notices. The telephone facility shall be capable of receiving requests for Claim Forms, Opt-Out Forms, or the long-form Settlement Notice, and providing: (i) a listing of the Settlement Website to be established pursuant to section 9.7; (ii) general information concerning deadlines for opting out of the Settlement Class, objecting to the Agreement, or filing a Claim Form; and (iii) the dates, times and locations of relevant proceedings, including the Final Approval Hearings. The telephone facility must be available in English and French. The toll-free number shall be maintained until the expiration of the last of the Applicable Warranty Periods. The toll-free number can be an existing number used by the Defendant.

- 9.7 No later than the publication of the first Settlement Notice pursuant to section 9.2, the Defendant shall cause an internet website concerning the Agreement to be established (the "Settlement Website"). The internet address of the Settlement Website shall be included in the Settlement Notices. The Settlement Website shall provide: (i) general information concerning deadlines for opting out of the Settlement Class, objecting to the Agreement, or filing a Claim Form; (ii) the dates, times and locations of relevant Court proceedings, including the Final Approval Hearings; (iii) a listing of the toll-free phone number to be established pursuant to section 9.6; (iv) copies of this Agreement, the long-form Settlement Notice, the Claim Form, and the Opt-Out Form; and (v) information concerning the submission of Claim Forms. The Settlement Website shall be maintained until the expiration of the last of the Applicable Warranty Periods. A similar website shall be available in the French language. The Settlement Website can be a webpage within the Defendant's website, currently www.bpcan.com.
- 9.8 The costs associated with establishing and maintaining the Settlement Website and toll-free telephone facility shall be paid by the Defendant.
- 9.9 The Defendant shall include in the section of its website concerning warranty claims for roofing shingles (in English and French): (i) a statement to the effect that a settlement has been entered into in relation to BP Organic Shingles; and (ii) a link to the Settlement Website. Such information shall be maintained on the Defendant's website until the expiration of the last of the Applicable Warranty Periods.
- 9.10 All Settlement Notices, Claim Forms and Opt-Out Forms disseminated in the Province of Quebec or sent to residents of Quebec, shall be available in both the English and French languages, or, where appearing in a publication that is in either the English or French language, shall appear in the language of that publication.

10. <u>SETTLEMENT CLASS MEMBERS' RIGHT TO OPT OUT AND DISMISSAL OF</u> OTHER PROCEEDINGS

- 10.1 A Settlement Class Member may opt out of or request exclusion from the Settlement Class. To exercise this right, members of the Canadian National Class must submit a fully completed Opt-Out Form to Canadian Class Counsel at an address to be specified in the Opt-Out Form postmarked no later than the Opt-Out Deadline, members of the Quebec Class must submit a fully completed Opt-Out Form to Canadian Class Counsel and the Quebec Court at addresses to be specified in the Opt-Out Form postmarked no later than the Opt-Out Deadline, and members of the U.S. Class must submit a fully completed Opt-Out Form to U.S. Class Counsel and the U.S. Court at addresses to be specified in the Opt-Out Form postmarked no later than the Opt-Out Deadline. Opt-Out Forms sent to an incorrect address or postmarked after the Opt-Out Deadline shall not be valid.
- 10.2 In seeking the Preliminary Approval Orders, the Plaintiffs and the Defendant will request that the Opt-Out Deadline be set on a date that is at least 60 days after the initial dissemination of the summary Settlement Notice.
- 10.3 The Opt-Out Form shall require the signature of the Settlement Class Member and any counsel who the Settlement Class Member has retained in respect of this litigation. The Opt-Out Form shall also require the Settlement Class Member to state the address of the property(ies) of the Settlement Class Member that might contain BP Organic Shingles, and specify the number of units of residential property or other structures at each address containing BP Organic Shingles owned by the Settlement Class Member opting-out of the settlement.
- 10.4 Settlement Class Members who have validly and timely opted out of the Settlement Class shall not be entitled to relief under or be affected by this Agreement.

- 10.5 Settlement Class Members who have opted out of the Settlement Class can withdraw their opt-out requests prior to the Effective Date, but only if they accept the benefits and terms of this Agreement and dismiss without costs and with prejudice any pending Other Action against the Released Parties.
- 10.6 Within seven (7) days after the Opt-Out Deadline, Class Counsel shall provide counsel for the Defendant, by electronic mail, facsimile, and/or hand delivery, with a list identifying each person who has submitted a complete and timely Opt-Out Form and attaching copies of all such Opt-Out Forms.
- 10.7 In the sole discretion of the Defendant, this Agreement may be unilaterally terminated if the number of Settlement Class Members opting out reaches a level that, in the Defendant's judgment, threatens to frustrate the essential purpose of this Agreement. The Defendant shall advise Class Counsel and the Courts, in writing, of this election within thirty (30) days of receiving the list of opt outs pursuant to section 10.6.
- 10.8 Each member of the Canadian National Class and the U.S. Class, who does not opt-out, shall be deemed to consent to the dismissal, without costs and with prejudice, of his, her or its Other Actions, to the extent that the Other Actions relate to claims released pursuant to section 15.1, against the Released Parties.
- 10.9 All Other Actions, to the extent that the Other Actions relate to claims released pursuant to section 15.1, that were commenced in Canada (except Quebec) or the United States by any Settlement Class Member who does not opt out shall be dismissed against the Released Parties without costs and with prejudice.
- 10.10 Each member of the Quebec Settlement Class who makes a claim under this Settlement Agreement shall be deemed to irrevocably consent to the dismissal, without costs and

without reservation, of his, her or its Other Actions, to the extent that the Other Actions relate to claims released pursuant to section 15.1, against the Released Parties.

10.11 Each Other Action, to the extent that the Other Actions relate to claims released pursuant to section 15.1, commenced in Quebec by a member of the Quebec Settlement Class who makes a claim under this Settlement Agreement shall be dismissed against the Released Parties, without costs and without reservation.

11. SETTLEMENT CLASS MEMBERS' RIGHT TO OBJECT

- 11.1 A Settlement Class Member may object to the Agreement. To exercise this right, members of the Canadian National Class or Quebec Class must submit written notification of the objections to Canadian Class Counsel at an address to be specified in the long-form Settlement Notice postmarked no later than the relevant Objection Deadline, and members of the U.S. Class must submit written notification of the objections to U.S. Class Counsel and the U.S. Court at the addresses to be specified in the long-form Settlement Notice postmarked no later than the relevant Objection Deadline. Objections sent to an incorrect address shall not be considered. Forthwith upon receiving an objection, Class Counsel shall forward a copy of such written objection to counsel for the Defendant.
- 11.2 To be valid and therefore considered by the appropriate Court, the written objection must bear the signature of the Settlement Class Member and must specify: (i) the Settlement Class Member's current address and telephone number; (ii) the address of the property(ies) that may contain BP Organic Shingles; (iii) the number of units of residential property or other structures at each address owned by the Settlement Class Member containing BP Organic Shingles; (iv) the exact nature of the objection; and (v) whether or not the Settlement

Class Member intends to appear at the appropriate Final Approval Hearing. If the Settlement Class Member is represented by counsel, the objection shall also be signed by counsel.

- 11.3 Objections that are sent to an incorrect address or do not contain the required information shall not be valid and will not be considered by the Courts.
- 11.4 In seeking the Preliminary Approval Orders, the Plaintiffs and the Defendant will request that the Objection Deadline be set on a date that is at least 60 days after the initial dissemination of the summary Settlement Notice.

12. FINAL APPROVAL ORDERS

- 12.1 This Agreement is subject to and conditioned upon the Courts granting final approval of the settlement embodied in this Agreement, and entering orders approving the Agreement and dismissing the Actions in accordance with the terms of this Agreement (the "Final Approval Orders").
- 12.2 The motions for final approval of this Agreement shall be scheduled sequentially, first in the Ontario Court, then in the Quebec Court and then in the U.S. Court.
- 12.3 If the Canadian Courts grant final approval of the Agreement, the Canadian Final Approval Orders shall, among other things:
 - (a) Approve finally this Agreement and its terms as being a fair, reasonable and in the best interests of Settlement Class Members within the meaning of section 29 of the Class Proceedings Act, 1992, S.O. 1992, c. 6 (or, in the case of Quebec, approve and homologate the Agreement as a transaction pursuant to section 1025 of the Quebec Code of Civil Procedure) and direct that the Agreement be implemented in accordance with its terms;
 - (b) Dismiss the relevant Canadian Action; and
 - (c) Provide that the applicable Court shall not make any order or give any direction in respect of any matter unless that order is conditional upon a

complementary order or direction being made or given by the other Courts with which it shares jurisdiction over that matter, as applicable.

- 12.4 If the U.S. Court grants final approval of the Agreement, the Final Approval Order shall, among other things:
 - (a) Approve finally this Agreement and its terms as being a fair, reasonable and adequate settlement as to the Settlement Class Members within the meaning of V.R.C.P. 23, the *Class Action Fairness Act*, and other applicable law, and direct that the Agreement be implemented in accordance with its terms;
 - (b) Dismiss the U.S. Action;
 - (c) Provide that the form and manner of notice given to the Settlement Class Members fairly and adequately informed them of all material elements of the U.S. Action and the proposed Agreement and constituted sufficient notice to the Settlement Class Members in accordance with V.R.C.P. 23 and due process requirements; and
 - (d) Provide that the U.S. Court shall not make any order or give any direction in respect of any matter unless that order is conditional upon a complementary order or direction being made or given by the other Courts with which it shares jurisdiction over that matter.
- 12.5 Prior to and as a condition precedent to filing with the Courts, Class Counsel shall obtain the signed, written approval of counsel for the Defendant as to the form and content of any motions materials and draft orders filed with the Courts in respect of the Final Approval Hearings.

13. **DEFENDANT'S RIGHT OF TERMINATION**

13.1 The Defendant's willingness to settle on a class-action basis and not to contest the accompanying certification of the Settlement Classes is dependent upon achieving finality and the desire to avoid the expense of this and other related litigation. Consequently, the Defendant has the unilateral right to terminate this Agreement by delivering a written notice of termination to Class Counsel if any of the following conditions subsequently occurs:

- (a) The Courts do not enter the Preliminary Approval Orders sought pursuant to section 8.1 or the Courts enter Preliminary Approval Orders in a materially modified form from that which was agreed upon by the Defendant in accordance with section 8.2. Materiality shall be determined by the Defendant, in its sole discretion, acting reasonably;
- (b) The Courts do not enter the Final Approval Orders sought pursuant to section 12 or the Courts enter Final Approval Orders in a materially modified form from that which was agreed upon by the Defendant in accordance with section 12.5. Materiality shall be determined by the Defendant, in its sole discretion, acting reasonably;
- (c) Any Preliminary Approval Order or Final Approval Order is not upheld on appeal, including review by any appellate court in the United States or Canada, or such Order is approved by any appellate court in a materially modified form from that which was agreed upon by the Defendant in accordance with section 12.5. Materiality shall be determined by the Defendant, in its sole discretion, acting reasonably;
- (d) The Effective Date has not occurred by December 31, 2012; or
- (e) The Defendant exercises its option under section 10.7 above to terminate the Agreement on account of opt-outs.
- 13.2 In the event that this Agreement does not become effective or is terminated for any reason:
 - (a) the provisional certification of the Settlement Classes shall cease;
 - (b) this Agreement shall become null and void and of no further force and effect;
 - (c) the Agreement and all negotiations, statements, communications, proceedings, and documents relating thereto, and the fact that the Parties agreed to the Agreement, shall be without prejudice to the rights of any Settling Party and shall not be used for any purpose whatsoever in any subsequent proceeding in the Actions or in any other action in any court or tribunal, and shall not be construed as an admission or concession by any Party of any fact, matter, allegation, or proposition of law; and
 - (d) the Settling Parties shall be restored without prejudice to their respective positions as if the Agreement had not been negotiated, made, or filed with any of the Courts, including but not limited to reservation of defenses including improper service and lack of personal jurisdiction.

13.3 If this Agreement is terminated or otherwise fails to take effect for any reason, the provisions of sections 3.5, 13.2 and 17.1 shall survive the termination and continue in full force and effect. The definitions and Schedules shall survive only for the limited purpose of the interpretation of sections 3.5, 13.2 and 17.1 within the meaning of this Agreement, but for no other purposes. All other provisions of this Agreement and all other obligations pursuant to this Agreement shall cease immediately.

14. <u>EFFECTIVE DATE</u>

- 14.1 The Agreement and the obligations of the Parties under this Agreement shall not become effective until, and are expressly conditioned upon, the occurrence of the Effective Date.
- Orders and the Final Approval Orders have become final. The Final Approval Orders shall become final when: (i) all periods within which to file an appeal from the Final Approval Orders have expired without the filing of any appeals; or (ii) in the event that an appeal from the Final Approval Orders of the Approval Orders is filed, a final order has been entered disposing of the appeal, and any time for further appeal has expired.

15. RELEASE

opted out of the Settlement Class pursuant to the terms of this Agreement shall be conclusively deemed to have released and forever discharged (as by an instrument under seal without further act by any person, and upon good and sufficient consideration), on behalf of themselves and their agents, heirs, executors and administrators, successors, attorneys, representatives, and assigns (collectively, the "Releasing Parties"), each of the Defendant, and its predecessors, successors, present or former parents or subsidiaries, affiliates, related parties, officers, directors, employees,

agents, attorneys, representatives, suppliers, distributors, vendors and assigns, which for greater certainty includes, without limitation, Emco Corporation, Emco Limited and BPCO Corp. (collectively, the "Released Parties"), from each and every claim of liability, including relief under federal law or the law of any province, territory or state, which arises out of Damage to BP Organic Shingles, including without limitation all claims or liability on account of or related to Damage to BP Organic Shingles, which were alleged or could have been alleged in the Actions and all claims (whether arising prior to the Effective Date or thereafter) for penalties, punitive damages, exemplary damages, statutory damages, damages based upon a multiplication of compensatory damages, court costs, or attorneys' fees or expenses, which might otherwise have been made in connection with any claim relating to Damaged BP Organic Shingles. Releasing Parties specifically reserve any and all other claims and causes of action relating to personal injury and/or damage to the interior of the house or other structure caused by the Damaged BP Organic Shingles. In addition, the Releasing Parties specifically reserve any and all other claims and causes of action against any and all other persons or entities who are not Released Parties. The Releasing Parties acknowledge and agree that such reservation creates no basis for a claim of indemnification or contribution, however denominated, by the non-party against the Released Parties, as Releasing Parties have released all claims on which liability could be found against the Released Parties, and is solely intended to preserve a Releasing Party's ability to seek relief against the non-party. This release shall apply to all related subrogation claims of the Settlement Class Members' subrogees or insurance carriers.

15.2 It is the intent of the Parties that no Releasing Party shall recover, directly or indirectly, any sums for claims released by operation of this Agreement from the Released Parties, other than sums received under this Agreement and that the Released Parties shall have

no obligation to make any payments to any non-parties for liability arising out of claims released by operation of this Agreement.

- (a) Releasing Parties agree that in any action brought by a Releasing Party against any non-party arising out of or related to Damage to BP Organic Shingles, should any such non-party file a claim against any Released Party for contribution or indemnification, however denominated, arising out of or related to Damage to BP Organic Shingles, the Releasing Parties shall reduce any judgment against the non-party by the percentage, amount, or share necessary under applicable law to fully discharge and relieve the Released Parties of liability to the non-party for claims for contribution and indemnification, however denominated.
- (b) The Releasing Parties agree that the provisions of this Agreement and any claim thereunder constitute a good faith settlement under California Civil Code §§ 877 and 877.6 and comparable laws in other states, that Class Counsel and the Releasing Parties shall cooperate fully in any effort of the Released Parties to establish such good faith settlement before any court (including, without limitation, by joining in any motion or other procedure and providing declarations and other evidence to establish such good faith settlement where requested by any Released Party) and that all payments made under this Agreement relate to claims arising out of or related to BP Organic Shingles.
- (c) If notwithstanding the intention of the Parties expressed therein, any release given by the Releasing Parties is not given its full effect by operation of law, then the Releasing Parties shall be deemed to have and do hereby transfer and assign to Released Parties all claims, if any, that were deemed not released, to the extent necessary to effectuate the intent of the release.
- (d) Class Counsel shall cooperate with Released Parties to ensure that the releases set forth in this section are given their full force and effect and that Releasing Parties comply with their obligations set forth in this Agreement.
- 15.3 In the event that any Releasing Party seeks to invoke California Civil Code § 1542, which provides that "a general release does not extend to claims which the creditor does not know or suspect to exist in his favor at the time of executing the release, which if known to him must have materially affected his settlement with the debtor" (or any other like provision of law) in connection with BP Organic Shingles, the Releasing Parties and each of them now

expressly waive the provision of California Civil Code § 1542 (and all other like provisions of law) to the full extent that these provisions may be applicable to this release. Each of the Releasing Parties hereby does, and shall be deemed to, assume the risk that facts additional, different, or contrary to the facts, which each believes or understands to exist, may now exist or may be discovered after this Agreement becomes effective. Each of the Releasing Parties agrees that any such additional, different, or contrary facts shall in no way limit, waive, or reduce the foregoing release, which shall remain in full force and effect.

15.4 It shall be a condition of receipt of settlement benefits under this Agreement that each Settlement Class Member receiving settlement benefits execute a written release in favour of the Released Parties in respect of all claims released pursuant to section 15.1, in a form to be approved by the Defendant. Such written release shall be contained in the body of the Claim Form to be filed by Settlement Class Members.

16. EXCLUSIVE REMEDY; DISMISSAL OF ACTION; JURISDICTION OF COURT

- 16.1 Settlement Class Members who have not validly and timely opted out of the Settlement Class, shall be:
 - (a) deemed Settlement Class Members for all purposes under this Agreement;
 - (b) bound by the terms of this Agreement (including, without limitation, any and all releases);
 - (c) deemed to have submitted to the jurisdiction of the Court in which they are a member of the Settlement Class; and
 - (d) bound by any subsequent proceedings, orders, and judgment issued by the Court in the applicable Action.
- 16.2 Upon the Effective Date, the Releasing Parties shall be barred from commencing, initiating, continuing, maintaining, asserting, or prosecuting, either directly or indirectly, whether in Canada, the United States or elsewhere, on their own behalf or on behalf of any class or any

other person, any action, suit, cause of action, claim or demand against any Released Party in respect of the claims released pursuant to section 15.1 or any matter related thereto, except in accordance with this Agreement.

- 16.3 Upon the Effective Date, the Actions and all claims and allegations concerning BP Organic Shingles therein asserted by the Settlement Class Members will be dismissed without costs and with prejudice.
- 16.4 Each of the Courts shall retain exclusive jurisdiction over the Action commenced in its jurisdiction and the parties thereto.
- 16.5 The Parties agree that no Court shall make any order or give any direction in respect of any matter of shared jurisdiction unless that order or direction is conditional upon a complementary order or direction being made or given by the other Court(s) with which it shares jurisdiction over that matter.
- 16.6 Notwithstanding the above, the Parties agree that all disputes, claims, or controversies arising in connection with, pursuant to, or related to the implementation, administration, enforcement, or interpretation of the terms of this Agreement (including any action or proceeding to enforce this Agreement or secure damages for its breach) shall be finally resolved by the Ontario Court, or if the Ontario Court directs, by a referee appointed by the Ontario Court, and the Parties submit to the jurisdiction of the Ontario Court for purposes of implementing, administering, enforcing, and interpreting the settlement provided for in this Agreement. However, matters relating specifically to members of the Quebec Class or the U.S. Class shall be determined by the Quebec Court or U.S. Court, as appropriate. To the extent necessary, the referee appointed under this section shall have the authority to conduct a reference in accordance with the Ontario Rules of Civil Procedure. The Parties shall each bear their own

costs of such Court hearing or reference, unless the Ontario Court or referee in its, his or her discretion finds it reasonable to assess such costs solely to the Plaintiffs or the Defendant. The Plaintiffs and the Defendant shall each be responsible for one half of the fees and disbursements of the referee, as fixed by the Ontario Court.

16.7 In considering the reasonableness of any request made pursuant to the provisions of this Agreement, the Ontario Court or the referee shall weigh the burden and expense of complying with the request against the importance of the subject matter of the request, as well as the corresponding interest of the Defendant in achieving a full resolution and closure of the Actions.

17. OTHER TERMS AND CONDITIONS

- 17.1 The Plaintiffs, Class Counsel and the Defendant agree that, whether or not it is terminated, this Agreement and anything contained herein, and any and all negotiations, documents, discussions and proceedings associated with this Agreement, and any action taken to carry out this Agreement, shall not be referred to, offered as evidence or received in evidence in any present, pending or future civil, criminal or administrative action or proceeding, except in a proceeding to approve and/or enforce this Agreement, or to defend against the assertion of claims released pursuant to section 15.1, or as otherwise required by law or as provided in this Agreement.
- 17.2 The Defendant represents and warrants that: (i) it has all requisite corporate power and authority to execute, deliver, and perform this Agreement and to consummate the transactions contemplated hereby; (ii) the execution, delivery, and performance of this Agreement have been duly authorized by all necessary corporate action on the part of the Defendant; (iii) its signatories to the Agreement have full authority to sign on behalf of and to

bind the Defendant to its terms; and (iv) this Agreement has been duly and validly executed and delivered by the Defendant and constitutes its legal, valid, and binding obligation.

- 17.3 Plaintiffs, Defendant, and Class Counsel agree to cooperate fully in seeking court approval of this Agreement and to use their best efforts to effect the consummation of the settlement provided for herein. They further agree to execute all such additional documents as shall be reasonably necessary to carry out the provisions of this Agreement.
- 17.4 The undersigned counsel represent that they have been fully authorized to execute this Agreement on behalf of their respective clients.
- 17.5 This Agreement shall be binding upon and inure to the benefit of the Parties to this Agreement and to all members of the Settlement Class and their respective agents, heirs, executors, administrators, successors or assigns.
- 17.6 This Agreement and its exhibits constitute the entire agreement of the Parties with respect to the subject matter thereof. The settlement contemplated by this Agreement is not subject to any condition not expressly provided for herein, and there exist no collateral or oral agreements relating to the subject matter of the Agreement. In entering this Agreement, no Party is relying on any promise, inducement, or representation other than those set forth herein. Any agreement purporting to change or modify the terms of this Agreement or the exhibits hereto must be in writing and be signed by counsel for each of the Parties.
- 17.7 All of the exhibits attached hereto or referred to herein are incorporated as if fully set forth in the body of this Agreement.
- 17.8 The waiver by any Party to this Agreement of any breach of its terms shall not be deemed or construed to be a waiver of any other breach of this Agreement, whether prior, subsequent, or contemporaneous.

- 17.9 This Agreement may be executed in any number of counterparts, each of which shall be deemed to be an original. All counterparts shall constitute one Agreement, binding on all Parties hereto, regardless of whether all Parties are signatories to the same counterpart, but the Agreement will be without effect until and unless all Parties to this Agreement have executed a counterpart.
 - 17.10 This Agreement shall be governed by the laws of the Province of Ontario.
- 17.11 This Agreement has been the subject of negotiations and discussions among the undersigned, each of which has been represented and advised by competent counsel, so that any statute, case law, or rule of interpretation or construction that would or might cause any provision to be construed against the drafter of this Agreement shall have no force and effect.
- 17.12 Any headings, subheadings, or titles herein are used for purposes of convenience only and have no other legal force, meaning, or effect.
- 17.13 The Recitals and Schedules to this Agreement are material and integral parts hereof and are fully incorporated into, and form part of, this Agreement.
- 17.14 In the computation of time in this Agreement, except where a contrary intention appears,
 - (a) where there is a reference to a number of days between two events, they shall be counted by excluding the day on which the first event happens and including the day on which the second event happens, including all calendar days; and
 - (b) only in the case where the time for doing an act expires on a holiday (as defined by the Ontario *Rules of Civil Procedure*), the act may be done on the next day that is not a holiday.

WHEREFORE, the undersigned have executed this Agreement on the 18th day of January, 2012.

HARRISON PENSA LLP

Jonathon Foreman

Atforneys for Ontario Plaintiffs

SISKINDS LLP

Charles Wright

Canadian Attorney for Defendant

LAUZON BÉLANGER LESPÉRANCE

INC.

Eric Lafreniere

Attorneys for Quebec Plaintiffs

SEDGWICK LLP

Michael Tanenbaum

U.S. Attorney for Defendant

AUDET & PARTERS, LLP

Michael McShane

WHEREFORE, the undersigned have executed this Agreement on the 18th day of January, 2012.

HARRISON PENSA LLP

Jonathon Foreman
Attorneys for Ontario Plaintiffs

SISKINDS LLF

Charles Wright

Canadian Attorney for Defendant

LAUZON BÉLANGER LESPÉRANCE INC.

Eric Lafreniere
Attorneys for Quebec Plaintiffs

SEDGWICK LLP

Michael Tanenbaum

U.S. Attorney for Defendant

AUDET & PARTERS, LLP

Michael McShane

WHEREFORE, the undersigned have executed this Agreement on the 18th day of January, 2012.

HARRISON PENSA LLP

Jonathon Foreman
Attorneys for Ontario Plaintiffs

LAUZON BELANGER LESPERANCE

INC.

Eric Lafreniere

Attorneys for Quebec Plaintiffs

SISKINDS LLP

Charles Wright

Canadian Attorney for Defendant

SEDGWICK LLP

Michael Tanenbaum

U.S. Attorney for Defendant

AUDET & PARTERS, LLP

Michael McShane

WHEREFORE, the undersigned have executed this Agreement on the 18th day of January, 2012.

HARRISON PENSA LLP

Jonathon Foreman
Attorneys for Ontario Plaintiffs

SISKINDS LLP

Charles Wright

Canadian Attorney for Defendant

LAUZON BÉLANGER LESPÉRANCE

INC.

Eric Lafreniere
Attorneys for Quebec Plaintiffs

SEDGWICK LLP

Michael Tanenbaum

U.S. Attorney for Defendant

AUDET & PARTERS, LLP

Michael McShane

WHEREFORE, the undersigned have executed this Agreement on the 18th day of January, 2012.

HARRISON PENSA LLP

Jonathon Foreman Attorneys for Ontario Plaintiffs SISKINDS LLP

Charles Wright Canadian Attorney for Defendant

LAUZON BÉLANGER LESPÉRANCE

INC.

Eric Lafreniere

Attorneys for Quebec Plaintiffs

SEDGWICK LLP

Michael Tanenbaum

U.S. Attorney for Defendant

CUNED GILBERT & LADUCA, LLP charles

Michael McShane

Exhibit "A" – Applicable Warranty Periods

Shingle Name	Applicable Warranty Period
Citadel	Sold 1994-2008: 20 years
Eclipse	Sold 1995-2001: 30 years
	Sold 2002-2003: original purchaser
	of shingles: lifetime; subsequent
	property owner: 30 years
	Sold 2004-2010: original purchaser
	of shingles: lifetime; subsequent
	property owner: 35 years
Eclipse HR	Sold 1994-2000: 30 years
Eclipse LS	Sold 2004-2006; 35 years
Elegance	Sold 1991-1993; 30 years
	Sold 1994-1997: 35 years
Elegance II	Sold 1996-1997: 30 years
Esgard 20	Sold 1992-1994: 20 years
Esgard 25	Sold 1987-1994: 25 years
Esgard Pro-Standard	Sold 1995-2000: 25 years
Europa	Sold 2001-2006: 25 years
Mirage	Sold 1998-2007: 25 years
Pro-Standard	Sold 1995-2005: 25 years
Rampart	Sold 1992-1993: 20 years
	Sold 1994-2010: 25 years
Roofmaster	Sold 1992-1993: 20 years
	Sold 1994-2008: 25 years
Roofmaster Classic	Sold 1996-2004: 25 years
Roofmaster Plus	Sold 1991-1993: 25 years
	Sold 1994-1996: 30 years
Super Eclipse	Sold 1995-1997: 35 years
Super Lok	Sold 1991-1993: 25 years
	Sold 1994-1998: 30 years
Tite Lok	Sold 1992-1993: 20 years
	Sold 1994-2007: 25 years
Tite On	Sold 1994-2004: 20 years
Tradition	Sold 1991-1993: 25 years
	Sold 1994-2009: 30 years
Weather-Tite	Sold 2004-2007: 25 years

NOTICE OF BP ORGANIC SHINGLES SETTLEMENT

For qualifying owners of buildings upon which BP Organic Shingles have been installed.

What is the litigation about? In the litigation (Marvin Sherebrin et al. v. Building Products of Canada Corp., Ontario Superior Court of Justice, Court File No. 4367/11CP; Diane Fitzsimmons v. La Cie Materiaux de Construction BP Canada, Quebec Superior Court, Court File No. 500-06-00580-114; and Robert S. Melillo et al. v. Building Products of Canada Corp., Vermont Superior Court, Chittenden Unit, Court File No. s618-11 cnc), it is alleged that BP Organic Shingles are subject to premature failure. Building Products of Canada Corp./La Cie Materiaux de Construction BP Canada (the "Defendant") denies these allegations and asserts that the BP Organic Shingles are free of any defect and will last the entire warranty period. However, in order to achieve certainty in terms of its ongoing obligations, the Defendant agreed to a settlement that will resolve the litigation. The settlement represents a voluntary resolution of claims. The Defendant does not admit any wrongdoing or liability.

The Ontario, Quebec and Vermont Courts will hold hearings to decide if the proposed settlement is fair, reasonable and adequate. In Ontario, the hearing will be on [date] at [time] in the City of London. In Quebec, the hearing will be on [date] at [time] in the City of Quebec. In Vermont, the hearing will be on [date] at [time] in the City of Burlington. The full addresses of the Courts are available online at [insert website].

More information about the litigation and the settlement (including copies of the settlement agreement and long-form Settlement Notice) is available online at [insert website].

What organic shingles are the subject of the settlement? The organic shingles that are the subject of the litigation are shingles made with a felt reinforcement base material that is saturated with asphalt, also known as "organic" roofing shingles, manufactured during the period from 1985 to 2010 under the brand names: Eclipse, Eclipse H/R, Eclipse LS, Super Eclipse, Weather-Tite, Mirage, Rampart, Tradition, Tite-Lok, Esgard Pro-Standard, Pro-Standard, Esgard 20, Esgard 25, Citadel, Tite-On, Roofmaster, Roofmaster Classic, Roofmaster Plus, Elegance, Elegance II, Europa, and Super Lok ("BP Organic Shingles").

Who is affected by the settlement? You might be affected by this settlement if you own or owned a building in Canada or the United States whose roofs contain or contained BP Organic Shingles.

You will only qualify for compensation under the settlement if: (1) you are a class member; (2) you submit a timely and complete Claim Form, along with the other required information; (3) you have not already removed the BP Organic Shingles; (4) you have experienced damage to your BP Organic Shingles, as set out in the settlement; and (5) the damage to the BP Organic Shingles was not caused by improper installation, improper maintenance, weather-related events, or some other factor unrelated to the manufacturing process, as set out in the settlement.

What are the settlement terms? The settlement codifies the BP Organic Shingles warranty and provides for some additional enhancements to the BP Organic Shingles warranty. The amount of compensation payable to qualifying claimants will depend on the following factors: (1) the terms of the applicable warranty; (2) the number of damaged BP Organic Shingles on the relevant roof slope; (3) the length of time the damaged BP Organic Shingles have been installed on the roof; and (4) whether the claimant selects the cash settlement option or the repairs settlement option.

The Defendant has agreed to pay class counsel fees plus reasonable expenses and costs in an amount not to exceed Cdn \$2,400,000.00. Class counsel fees are subject to the approval of the Vermont Court.

If I am affected by the settlement, what are my options?

Exclude yourself. If you exclude yourself or "opt-out," you will not be eligible for compensation under the settlement, but you will be free to sue the Defendant on your own about the claims discussed in this Notice. For instructions on excluding yourself, see the long-form Settlement

Notice, available online at [website]. The deadline for opting out is [DATE].

Make Submissions. If wish to make submissions to the appropriate Court regarding the settlement, you can file written submissions or ask to make oral submissions. For instructions on making written or oral submissions, see the long-form Settlement Notice, available online at [website]. The deadline for making written submissions or applying to make oral submissions is [date].

File a Claim Form. To be eligible for compensation under the settlement, you must submit a timely and complete Claim Form. Claim Forms can be obtained online at [website] or by calling toll-free [phone number]. See the long-form Settlement Notice for details regarding the claims filing deadline.

How do I obtain more information? More information about your rights under the proposed settlement, including copies of the settlement agreement and the long-form Settlement Notice, is available online at [website] or by calling [toll-free number]. PLEASE DO NOT CALL THE COURTS.

ONTARIO SUPERIOR COURT OF JUSTICE

Marvin Sherebrin et al. v.

Court File No. 4367/11CP

Building Products of Canada Corp.

SUPERIOR COURT OF QUÉBEC

Diane Fitzsimmons v. La Cie Materiaux de Construction BP Canada

Court File No. 500-06-00580-114

VERMONT SUPERIOR COURT

Robert S. Melillo et al. v. Building Products of Canada Corp.

Court File No. s618-11 cnc

NOTICE OF BP ORGANIC SHINGLES SETTLEMENT

Canadian National Class Hearing Date & Tin	ne:	,	_, at	a.m.
Québec Class Hearing Date & Time:	,	, at _	a.m.	
U.S. Class Hearing Date & Time:		. at	a.m.	

BE SURE TO CHECK THE SETTLEMENT WEBSITE [WEBSITE] FOR REGULAR UPDATES. YOU HAVE A DUTY TO KEEP YOURSELF INFORMED.

This Notice was authorized by the Courts. It is not from a lawyer. You are not being sued.

TO: ALL PERSONS THAT OWN OR OWNED HOMES, RESIDENCES, BUILDINGS OR OTHER STRUCTURES LOCATED IN CANADA OR THE UNITED STATES WHOSE ROOFS CONTAIN OR CONTAINED BP ORGANIC SHINGLES ("CLASS MEMBERS"). YOUR LEGAL RIGHTS ARE AFFECTED BY A CLASS ACTION SETTLEMENT. PLEASE READ THIS ENTIRE NOTICE CAREFULLY.

The purpose of this Notice is to advise Class Members of the proposed settlement of the above-noted class action lawsuits and Class Members' rights and options with respect to the proposed settlement. A copy of the proposed settlement is available online at [website]. This Notice is not an opinion of the Courts about the merits of the claims or defenses of the parties in the lawsuits. Building Products of Canada Corp./La Cie Materiaux de Construction BP Canada (the "Defendant") denies the allegations asserted in the lawsuits, but has agreed to enter into the proposed settlement in order to achieve certainty in terms of its ongoing obligations. The settlement represents a voluntary resolution of claims. The Defendant does not admit any wrongdoing or liability. The Defendant asserts that the BP Organic Shingles are free of any defect and, if properly installed, will last throughout the entire warranty period.

The Courts have issued orders approving this Notice and certifying the Classes for settlement purposes, and the Vermont Court has granted preliminary approval of the settlement. Hearings will be held before the Courts to consider the fairness, adequacy, and reasonableness of the settlement.

YOUR LEGAL RIGHTS AND CHOICES		POST MARK ON OR BEFORE
Exclude yourself	If you exclude yourself or "opt out," you will not be	[date]

	eligible for compensation under the settlement and will not be able to make submissions on the settlement. You will not be bound by the settlement or any related court orders, but you will be free to sue the Defendant on your own about the claims discussed in this Notice. (see No. 19 below)	
Make written submissions	You can submit written submissions to the appropriate Court, raising any comments or concerns you might have about the proposed settlement. (see No. 20 below)	[date]
Make oral submissions	You can ask to speak to the appropriate Court, raising any comments or concerns you might have about the proposed settlement. (see No. 20 below)	[date]
Submit a Claim Form To be eligible for compensation under the settlement, you must submit a timely and complete Claim Form. (see No. 14 below)		See No. 15

BASIC INFORMATION

1. Why was this Notice sent to me?

This Notice is published pursuant to Orders by the Courts. You received this Notice either because the Defendant's records indicate that you might be a Class Member or you requested that a copy of this Notice be sent to you. If you are a tenant or occupant of a building at the address this Notice was sent, please ensure that a copy of the Notice is provided to the landlord or owner of the building. You have legal rights and choices to make before the Courts decide whether to finally approve the settlement. This Notice summarizes:

- What the lawsuits are about.
- Who is affected by the settlement.
- The basic terms of the settlement.
- Your legal rights with respect to the settlement.

You are encouraged to review the complete settlement, a copy of which is available online at [website]. This notice does not alter the settlement.

2. What is a class action?

In a class action lawsuit, one or more people called "representative plaintiffs" sue on behalf of other people who are similarly situated. This group of people together is referred to as the "class" or "class members." One court decides all the issues in the lawsuit for all class members, except for those who exclude themselves from the class.

In this case, the Ontario Court will determine the issues for all class members in Canada (excluding Quebec); the Quebec Court will determine the issues for all class members in Quebec; and the Vermont Court will determine the issues for all class members in the United States.

3. What is this class action lawsuit about?

In this class action lawsuit, it is alleged that BP Organic Shingles are defective and might fail prematurely, resulting in possible damage to the home, residence, building or other structure. BP Organic Shingles were manufactured and distributed by or on behalf of the Defendant, and installed on homes, residences, buildings and other structures in Canada and the United States. The lawsuits do not relate to

damages for personal injury or consequential damages to the interior of the building, and the settlement does not affect claims for such damages. The Defendant denies the allegations asserted in the lawsuits and asserts that the BP Organic Shingles are free of any defect and, if properly installed, will last throughout the entire warranty period. However, in order to achieve certainty in terms of its ongoing obligations, the Defendant agreed to a settlement that will resolve the claims at issue in the class action lawsuit. The settlement represents a voluntary resolution of claims.

The class action lawsuit involves three separate proceedings in Ontario, Quebec and the United States. The Ontario action is being heard by Justice [name] of the Ontario Superior Court of Justice and is entitled Marvin Sherebrin et al. v. Building Products of Canada Corp., Court File No. 4367/11CP.

The Quebec action is being heard by Justice [name] of the Quebec Superior Court and is entitled *Diane Fitzsimmons v. La Cie Materiaux de Construction BP Canada*, Court File No. 500-06-00580-114.

The U.S. action is being heard by Judge [name] of the Vermont Superior Court, Chittenden Unit and is entitled Robert S. Melillo et al. v. Building Products of Canada Corp., Court File No. s618-11 cnc.

4. What are the shingles that are the subject of this class action lawsuit?

The shingles that are the subject of this class action lawsuit (referred to in this Notice as "BP Organic Shingles") are shingles made with a felt reinforcement base material that is saturated with asphalt, also known as "organic" roofing shingles, manufactured during the period from 1985 to 2010 under the brand names: Eclipse, Eclipse H/R, Eclipse LS, Super Eclipse, Weather-Tite, Mirage, Rampart, Tradition, Tite-Lok, Esgard Pro-Standard, Pro-Standard, Esgard 20, Esgard 25, Citadel, Tite-On, Roofmaster, Roofmaster Classic, Roofmaster Plus, Elegance, Elegance II, Europa, and Super Lok.

5. Why is the class action lawsuit being settled?

The Courts have not decided in favour of the Class or the Defendant. The Defendant denies any fault, wrongdoing, illegal conduct, or liability whatsoever on its part, and asserts that the BP Organic Shingles are free of any defect and, if properly installed, will last the entire warranty period. However, in order to achieve certainty in terms of its ongoing obligations, the Defendant agreed to a settlement that will resolve the claims at issue in the class action lawsuit. The settlement represents a voluntary resolution of claims. The Defendant does not admit any wrongdoing or liability.

Both sides have agreed to a settlement in order to avoid the risks and costs associated with trial. It could take several years before a trial is heard and decided, and any subsequent appeals resolved.

WHO IS AFFECTED BY THE SETTLEMENT?

6. Who is affected by the settlement?

The Courts have certified the class actions for settlement purposes. You are affected by the settlement if you fall within the scope of the certified classes.

In Canada:

The Ontario Superior Court of Justice certified a Canadian National Class for settlement purposes only as follows:

All individuals and entities, that own or owned, homes, residences, buildings, or other structures located in Canada whose roofs contain or contained BP Organic Shingles, excluding members of the Quebec Class.

For certain persons in the Province of Québec:

The Superior Court of Québec authorized a Québec Class for settlement purposes only as follows:

All natural persons, as well as all legal persons established for a private interest, partnerships and associations having no more than 50 persons bound to it by contract of employment under its direction or control during the 12-month period preceding September 28, 2011, that own or owned, homes, residences, buildings, or other structures located in the Province of Quebec whose roofs contain or contained BP Organic Shingles.

Note: If you own or owned homes, residences, buildings, or other structures located in the Province of Quebec and you have more than 50 employees or are a government entity or legal person established in the public interest, you cannot be part of the Québec Class, but you are part of the Canadian National Class.

In the United States:

The Vermont Superior Court, Chittenden Unit, certified a U.S. Class for settlement purposes only as follows:

All individuals and entities, that own or owned, homes, residences, buildings, or other structures located in the United States whose roofs contain or contained BP Organic Shingles.

For purposes of this Notice, the Canadian National Class, the Québec Class and the U.S. Class are collectively referred to as the "Class".

7. Are there any exclusions from the Class?

You are not a class member even if you have or had BP Organic Shingles installed on your home, residence, building or other structure if:

- You excluded yourself from the settlement (see No. 19 below).
- You previously filed a claim concerning your BP Organic Shingles in any court of law and your claim was resolved by a final judgment or settlement, whether or not that judgment was favourable to you.
- You are an entity related to the Defendant.

If you have previously filed a warranty claim with the Defendant, please see No. 17 for more information regarding your rights under the settlement.

8. How do I know if I have BP Organic Shingles?

There are several ways to determine if you have BP Organic Shingles:

- Check your receipts, invoices, brochures, etc. from when you purchased your shingles.
- Contact the contractor or company that installed your shingles.
- Ask an experienced roofer.
- Go to the website [website] for further information and photographs.

9. Who is eligible for compensation under the settlement?

Generally speaking, you will be eligible for compensation under the settlement if you meet the following criteria:

• You are a class member.

- You submit a timely and complete Claim Form, along with any other required information.
- You have not already removed the BP Organic Shingles.
- You have experienced damage to your BP Organic Shingles. Such damage includes blistering, clawing, cracking, curling, cold weather curling, fishmouthing, and delamination. See section 4.5 of the settlement for complete information about the type of damage covered by the settlement.
- The damage to the BP Organic Shingles was not caused by improper installation, improper maintenance, weather-related events, or some other factor unrelated to the manufacturing process. See section 4.6 of the settlement for complete information.

BASIC TERMS OF THE SETTLEMENT

10. What disputes does the settlement resolve?

The proposed settlement seeks to resolve all past, present and future, known and unknown, direct or contingent claims of Class Members against the Defendant and its related entities, including Emco Corporation and BPCO Corp., in relation to damage to BP Organic Shingles. The settlement does not affect Class Member claims for personal injury or consequential damage to the interior of the building.

11. How does the settlement work?

The settlement codifies the BP Organic Shingles warranty and provides for some additional enhancements to the BP Organic Shingles warranty. You will only be eligible for compensation if you meet the criteria set out in No. 9 above. Assuming you are eligible for compensation, the amount of the compensation payable to you will depend on the following factors:

- The terms of the applicable warranty.
- The number of damaged BP Organic Shingles on the relevant roof slope.
- The length of time the BP Organic Shingles have been installed on your roof.
- Whether the claimant selects the cash settlement option or the repairs settlement option.

12. What compensation is payable under the settlement?

Class Members can select between the cash settlement option and the repairs settlement option.

In reviewing the settlement options, it is useful to keep the following in mind:

- A "square" means 100 square feet of roofing shingles. A "square" is typically composed of three bundles/packages of shingles.
- The non-proration period is the period during which the Defendant is obligated to pay the full settlement benefits without any "proration" or discount to account for the years of use you have already received from your shingles. The non-proration period is the later of five years from the date of installation and the non-proration period as per the applicable BP warranty. The applicable non-proration periods are listed online at [website].
- After the non-proration period, settlement benefits will be prorated or discounted to account for the years of use you have already received from your shingles. For example, if you file a claim for BP Organic Shingles that have a warranty period of 30 years (360 months) exactly 12 years after they were installed, you would have used the shingles for 144 months and 216 months would remain on the warranty. Your settlement benefits would be calculated by multiplying the compensation amount by 216/360 (the proportion of the warranty that remains).

The number of squares on which settlement benefits will be calculated will depend on the size of the roof slope and the extent of damaged BP Organic Shingles on the roof slope. If the damaged BP Organic Shingles make up less than 5% of the roof slope, settlement benefits will be calculated based on the number of damaged BP Organic Shingles. If the damaged BP Organic Shingles make up more than 5%

of the roof slope, settlement benefits will be calculated based on the number of squares on the entire roof slope.

Under the cash settlement option, during the non-proration period, eligible Class Members in Canada will receive Cdn \$75/square and eligible Class Members in the United States will receive US \$75/square. After the non-proration period, the Cdn \$75/square (or US \$75/square, as appropriate) will be prorated or discounted to account for the use the eligible Class Members have already received from their BP Organic Shingles.

Under the repair settlement option, eligible Class Members will be compensated for the costs of repairing and replacing the damaged BP Organic Shingles, in accordance with the terms of the applicable warranty. During the non-proration period, the Defendant will reimburse the eligible Class Member for the full costs of repairs or replacement in accordance with the terms of the applicable warranty. After the non-proration period, the Defendant will reimburse the eligible Class Member for the costs of repairs or replacement in accordance with the terms of the applicable warranty on a prorated basis, to account for the use the eligible Class Members have already received from their BP Organic Shingles. In completing the repairs or replacement, eligible Class Members can select the roofer of their choice, but must use shingles manufactured by the Defendant. Copies of the applicable warranties are available online at [website].

13. What happens if the settlement is not approved by the Courts?

If the settlement is not approved by the Courts at the Final Approval Hearings, the settlement will terminate and all Class Members and Parties will be restored to the position they were in before the settlement was signed.

YOUR RIGHTS - SUBMITTING A CLAIM

14. How do I claim for settlement benefits?

In order to claim for settlement benefits, you must submit a completed Claim Form, together with any required information. You must also respond to reasonable requests from the Defendant for more information, including sample shingles.

Claim Forms can be obtained online at [website] or by calling toll-free [phone number].

15. What is the claims deadline?

Claim Forms must be submitted no later than 150 days after the later of the Effective Date of the settlement (which will be posted online at [website] once known) and the Claimant's discovery of the need for repairs or replacement and, in any event, no later than the expiry of the applicable warranty period and prior to the removal of the BP Organic Shingles from the home, residence, building or other structure upon which they were installed.

The following chart sets out the applicable warranty periods (the warranty periods for BP Organic Shingles sold prior to the dates specified below have already expired):

Shingle Name	Warranty Period
Citadel	Sold 1994-2008: 20 years
Eclipse	Sold 1995-2001: 30 years
	Sold 2002-2003: original purchaser
	of shingles: lifetime; subsequent
	property owner: 30 years
	Sold 2004-2010; original purchaser

Shingle Name	Warranty Period
	of shingles: lifetime; subsequent
	property owner: 35 years
Eclipse HR	Sold 1994-2000: 30 years
Eclipse LS	Sold 2004-2006: 35 years
Elegance	Sold 1991-1993: 30 years
	Sold 1994-1997: 35 years
Elegance II	Sold 1996-1997: 30 years
Esgard 20	Sold 1992-1994: 20 years
Esgard 25	Sold 1987-1994: 25 years
Esgard Pro-Standard	Sold 1995-2000: 25 years
Europa	Sold 2001-2006: 25 years
Mirage	Sold 1998-2007: 25 years
Pro-Standard	Sold 1995-2005: 25 years
Rampart	Sold 1992-1993: 20 years
	Sold 1994-2010: 25 years
Roofmaster	Sold 1992-1993: 20 years
	Sold 1994-2008: 25 years
Roofmaster Classic	Sold 1996-2004: 25 years
Roofmaster Plus	Sold 1991-1993: 25 years
	Sold 1994-1996: 30 years
Super Eclipse	Sold 1995-1997: 35 years
Super Lok	Sold 1991-1993: 25 years
	Sold 1994-1998: 30 years
Tite Lok	Sold 1992-1993: 20 years
	Sold 1994-2007: 25 years
Tite On	Sold 1994-2004: 20 years
Tradition	Sold 1991-1993: 25 years
	Sold 1994-2009: 30 years
Weather-Tite	Sold 2004-2007: 25 years

16. What happens if I bought the building with BP Organic Shingles already installed? Provided you can meet the other eligibility requirements and have not assigned the right to file a claim to the previous owner, you can file a claim.

17. What happens if I have previously filed a warranty claim with the Defendant?

If you previously filed a warranty claim in relation to your damaged BP Organic Shingles and your warranty claim was accepted, you cannot claim under the settlement with respect to the same damaged BP Organic Shingles, but you can claim under the settlement with respect to other damaged BP Organic Shingles.

If you previously filed a warranty claim in relation to your damaged BP Organic Shingles and your warranty claim was rejected after June 9, 2009, you can resubmit a claim under the settlement. If your warranty claim was rejected before June 9, 2009, you cannot claim under the settlement with respect to the same BP Organic Shingles unless the condition of the shingles has worsened such that the condition now qualifies as damage to BP Organic Shingles as per the terms of the settlement (see No. 9 above). You can however claim with respect to BP Organic Shingles that were not the subject of the previously-filed warranty claim.

18. What happens if I file a claim under the settlement and I believe that my claim was improperly denied?

If you believe that your claim under the settlement was improperly denied, you can appeal to an independent third-party. Instructions for appealing will be included in the claims decision.

YOUR RIGHTS - EXCLUDING YOURSELF

19. How do I exclude myself or opt out of the settlement?

You can exclude yourself or opt out of the settlement. If you exclude yourself, you will not be eligible for compensation under the settlement and cannot make submissions to the Courts regarding the settlement. You will not be bound by the settlement or any Court orders issued in the class action lawsuits. By excluding yourself, you keep any right to file or proceed with any claim you might have against the Defendant regarding damage to BP Organic Shingles. You do not need to exclude yourself in order to bring a claim you might have against the Defendant in respect of personal injury or consequential damages to the interior of the building.

If you are a member of the Canadian National Class or the U.S. Class and you have commenced individual litigation against the Defendant in relation to your BP Organic Shingles, your individual litigation will be dismissed unless you opt out. If you are a member of the Quebec Class and you have commenced individual litigation against the Defendant in relation to your BP Organic Shingles, your individual litigation will be dismissed if you file a claim under the settlement.

To exclude yourself, you must complete and send an Opt-Out Form postmarked no later than [date] via first class mail to the following addresses. A copy of the Opt Out Form is available online at [website].

Class Members with claims related to property(ies) in Canada (outside of the Province of Québec and those with buildings in the Province of Québec who employ more than 50 employees or were established in the public interest) should mail Opt-Out Forms to:

Jonathon Foreman Harrison Pensa LLP 450 Talbot Street P.O. Box 3237 London, ON N6A 4K3

For Class Members with claims related to property(ies) in the Province of Québec (except those who employ more than 50 people or were established in the public interest) should mail Opt-Out Forms to:

[name]

AND

Superior Court of Québec

Lauzon Bélanger Lespérance Inc. 286 Saint-Paul West, Suite 100

Montréal, QC H2Y 2A3

Palais de Justice 300, boul. Jean-Lesage Québec City, QC G1K 8K6

Class Members with claims related to property(ies) in the United States should mail Opt-Out Forms to:

US pl's counsel

AND

Vermont Superior Court Chittenden Unit 175 Main Street Burlington, VT 05401

YOUR RIGHTS - MAKING SUBMISSIONS

20. How do I make submission to the Courts regarding the Settlement?

If you are a member of the Class and you have comments or concerns regarding the settlement, you can make written or oral submissions to the appropriate Court. You can hire a lawyer at your own expense to act on your behalf in making submissions to the appropriate Court. To make submissions, you or your lawyer must submit an objection letter stating your comments or concerns regarding the settlement. Members with claims relating to buildings located in Canada may only make submissions before the applicable Canadian Court and appear at the appropriate Canadian Final Approval Hearing. Members with claims relating to buildings located in the United States may only make submissions before the Vermont Court and appear at the U.S. Final Approval Hearing. The Courts are entitled and empowered to approve the settlement in spite of any submissions.

The objection letter must include the following information:

- The name and title of the lawsuit:
 - o Members of the Canadian National Class (outside of the Province of Québec and those with buildings in the Province of Québec who employ more than 50 employees or were established in the public interest) should mention: *Marvin Sherebrin et al. v. Building Products of Canada Corp.*, Court File No. 4367/11CP.
 - o Members of the Class in the Province of Québec (except those who employ more than 50 people or were established in the public interest) should mention: *Diane Fitzsimmons v. La Cie Materiaux de Construction BP Canada*, Court File No. 500-06-00580-114.
 - o Members of the U.S. Class should mention: Robert S. Melillo et al. v. Building Products of Canada Corp., Court File No. s618-11 cnc.
- Your current address and telephone number.
- The address of the property(ies) that contains BP Organic Shingles.
- The number of units of residential property or other structures at each address containing BP Organic Shingles.
- The nature of your comments or concerns regarding the settlement.
- State whether you or your lawyer will appear at the appropriate Final Approval Hearing to speak on your comments or concerns.
- Provide copies of any documents that you or your lawyer wish to present at the appropriate Final Approval Hearing.
- Include your signature (even if you are represented by a lawyer), and the signature of your lawyer if you have retained one.

Your objection letter must be mailed and postmarked no later than [date] to the following addresses:

Class Members with claims related to property(ies) in Canada (outside of the Province of Québec and those with buildings in the Province of Québec who employ more than 50 employees or were established in the public interest) should mail the objection letter to:

Jonathon Foreman Harrison Pensa LLP 450 Talbot Street P.O. Box 3237 London, ON N6A 4K3

For Class Members with claims related to property(ies) in the Province of Québec (except those who employ more than 50 people or were established in the public interest) should mail the objection letter to:

[name]

Lauzon Bélanger Lespérance Inc. 286 Saint-Paul West, Suite 100 Montréal (Québec) H2Y 2A3

Class Members with claims related to property(ies) in the United States should mail the objection letter to:

US pl's counsel

AND

Vermont Superior Court

Chittenden Unit 175 Main Street Burlington, VT 05401

CLASS COUNSEL

21. Do I have a lawyer in the class action lawsuit?

The following firms act on behalf of the plaintiffs and Class Members in the class action lawsuit. They are collectively referred to as "Class Counsel". If you have any question regarding the settlement, you should contact the appropriate Class Counsel.

Counsel for Canadian Plaintiffs	Counsel for Québec Plaintiffs
Jonathon Foreman	[insert name]
Harrison Pensa LLP	Lauzon Bélanger Lespérance Inc.
450 Talbot Street	286 Saint-Paul West, Suite 100
P.O. Box 3237	Montréal (Québec) H2Y 2A3
London, ON N6A 4K3	Tel:
Toll free: 1-800-263-0489	Email:
Email: jforeman@harrisonpensa.com	
Co-lead counsel for U.S. Plaintiffs	Co-lead counsel for U.S. Plaintiffs
Charles J. LaDuca	Michael McShane
Cuneo Gilbert & LaDuca, LLP	Audet & Partners, LLP
507 C Street, NE	221 Main Street
Washington, DC 20002	Suite 1460
Tel: (202) 789-3960	San Francisco, CA 94105
Email: charles@cuneolaw.com	Toll free: 1-800-965-1461
	Email: mmcshane@audetlaw.com

22. How will these lawyers be paid?

You will not be individually billed or charged for Class Counsel. Under the terms of the settlement, the Defendant has agreed to pay Class Counsel's fees and expenses in an amount not to exceed Cdn \$2,400,000.00. Class Counsel fees are subject to the approval of the Vermont Court.

THE FINAL APPROVAL HEARINGS

23. When and where will the Courts decide whether to approve the settlement?

Final Approval Hearings will be held in Ontario, Quebec and the United States. At these hearings, the Courts will consider whether the settlement is fair and adequate. The Courts will consider any timely

written submissions and will listen to people who have asked to make oral submissions. After the hearings, the Courts will decide whether to approve the settlement.
The Final Approval Hearing for the Canadian National Class will be held on a.m. at Ontario Superior Court of Justice, Court House, 80 Dundas Street, London, ON N6A 6A3. The Final Approval Hearing for the Quebec Class will be held on at a.m. on at a.m. at Superior Court of Québec, Palais de Justice, 300, boul. Jean-Lesage, Québec City, Québec G1K 8K6.
The Final Approval Hearing for the U.S. Class will be held on at a.m. on at a.m. at Vermont Superior Court, Chittenden Unit, 175 Main Street, Burlington, Vermont, 05401.
24. Do I have to attend the Final Approval Hearings? No, you are not required to attend the Final Approval Hearings, but you are welcome to attend if you wish to. You may also pay your own lawyer to review the settlement or attend the appropriate Final Approval Hearing on your behalf.
25. May I speak at the Final Approval Hearings? You can participate in the Final Approval Hearing through written or oral submissions. See No. 20 above for instructions. ADDITIONAL INFORMATION
26. How can I obtain more information about the settlement? Additional information about the settlement, including a copy of the settlement, is available online at

You should regularly check the settlement website [website] for updates on the status of the litigation.

[website], by calling toll-free [number] or by contacting Class Counsel at the addresses listed above (see

No. 21).



NOTICE PLAN FOR THE PROPOSED BUILDING PRODUCTS OF CANADA SETTLEMENT

FEBRUARY, 2012
Confidential



Table of Contents

1.	Firm Overview	3
	Background	
	Objectives of the Notice Plan	
	Class Definition	
	Plan Overview	
6.	Review of Media Selection	7
7.	Pricing	13
8.	Measurements	13

1. Firm Overview

Our firm was established in London over fifteen years ago and comprises twelve partners, along with a combination of other supporting staff of Chartered Accountants, Certified General Accountants and students in training for each designation and program. In total, we have over 60 full-time employees, including Partners. Our firm competes in four areas: accounting services, business valuations, wealth management, and class action administration. Class action administration services are provided through NPT RicePoint Class Action Services Inc.

Claims Administration Services

NPT RicePoint Class Action Services grew out of a merger between NPT Administration Inc. and RicePoint Class Action Services Inc. Our combined firm offers a complete line of bi-lingual class action administration services for price-fixing, securities, product liability, and employment/pension settlements. These services address each phase of the settlement process, from designing custom websites & databases for claimants, to administering the Notification, Claims Processing and Settlement Distribution stages of a case. The NPT RicePoint team has administered 25 settlements with over \$90 million in settlement assets.

Notice Services

In addition to traditional claims administration, NPT RicePoint has a Notice Group that designs and executes all aspects of Notice including:

- Social Media Campaigns: To assist in pre certification case collection or post settlement notice plans.
- Notice Placement: Media estimating, negotiating, preparation and booking of orders, auditing and reporting.
- Notice Plan Design: Developing a strategy and tactics to effectively and efficiently reach class members. To establish a budget for use during settlement negotiations.
- Search: Locating missing class members.

2. Background

• Asphalt Shingle Market

Four out of every five homes are roofed with asphalt shingles (glass fiber and organic). In Canada, this represents approximately 500,000 homes a year according to the Canadian Asphalt Shingle Manufacturers Association ("CASMA"). Although no statistics are kept on the split between professional installations versus do-it-yourself installation, anecdotal evidence supplied by CASMA suggests that the vast majority of roofing contracts are completed by professionals.

In Canada, production of asphalt shingles is dominated by two large manufacturers, each with 40-45% share. Both market their products through building supply centres and big box retailers.

• Organic Shingles

Organic roof shingles refers to using cellulose (paper) as the substrate on which the shingle is constructed. The cardboard or cellulose shingle mat is impregnated with hot asphalt and coated with mineral granules to form a sunlight and weather resistant upper surface.

Glass fiber shingles have a glass fiber reinforcing mat manufactured to the shape of the shingle. This mat is then coated with asphalt which contains mineral fillers.

Potential Organic Shingle Defect

The settlement resolves a lawsuit over whether or not BP Canada organic roofing shingles are defective and failed to perform as promised when installed on homes in Canada and the North Eastern United States.

Notice Plan

There will be a single notice in connection with the Settlement. It will advise class members of the settlement approval hearings, the right to opt out, the right to object to the settlement, and the process for filing a claim.

3. Objectives of the Notice Plan

To reach the greatest number of Eligible Class Members so that they become aware of the settlement, their rights, and the claim process.

4. Class Definition

The class includes persons in Canada or the United States that own or owned a building whose roof contain or contained BP organic shingles.

The BP organic shingles at issue were sold between 1985 and 2010 under the brand names: Eclipse, Eclipse H/R, Eclipse LS, Super Eclipse, Weather-tite, Mirage, Rampart, Tradition, Tite-Lok, Esgard Pro-Standard, Esgard 20, Esgard 25, Citadel, Tite-On, Lo/Slope, Roofmaster, Roofmaster Classic, Roofmaster Plus, Elegance, Elegance II, Europa, and Super Lok.

5. Plan Overview

Considerations

- 1. The estimated class size is 150,000.
- 2. There are approximately 5,000 consumers who have previously filed warranty claims that are not yet resolved. Up to date address information is available for each of these potential class members.
- 3. The class is concentrated (refer to Table I):
 - Approximately, 96% of the sales over the past 25 years were made in Canada, of which close to 35% were in Quebec.
 - 4% of the sales were to the United States in states near the eastern border with Canada (Michigan, New York, Massachusetts etc.).

Table I - Sales Distribution

(graphic basel) at the Court	មានាំថា ប្រើបានឲ្យ		[154] 2 Made 1620	11: 1 / 1
Da estintian	H Hote	Percent	Arcanul	Pastqui
3.10 电气量 1.	51,998,363	35 36%	3,188,715	26 22%
1.48 / 12 / 10 v	42,204,041	28.70%	4:554:250	37/45%
$\Phi_{ij} = \Phi_{ij} \cap \Phi_{ij}$	15,364,459	10.45%	1,256,190	10/33%
Mary Books	7,121,409	4 84%	448,765	3 69%
myapha be milian kilikasi	5,559,744	3.78%	387/160	3:18%
Program Control	5,384,775	3.66%	376,830	3.10%
After the second second	4/724:030	3:21%	Control of the Contro	0.24%
object a Britisha.	4,196,081	2185%	197,245	1,62%
State of the Assessment	3,861,360	2.63%	1,642,715	13:51%
All the control	1;865,205	1.27%		
Part of Secretary Control	1,113,774	0 76%		
with the we	796 631	0.54%	24 Sec. 12 Sec. 2010	20.00
Chileton Awage to safe	757,506	0.52%	53,080	0.44%
Class Stad of the	682/901	0/46%	04.0 (2.8)	
postaloj. 1817. ili glasiona	558,745	0.38%		
and the first of the second of	313,269 188,629	0.21% 0.13%	potavijalno bisilik	
A Commence of the Commence of	128.103	0.09%		
	60.839	0.04%		
	38.381	0.03%	Control of the Contro	
Will design of the	32.7774	0.02%		
gradul force in the control	23,900	0.02%	14,220	0:12%
West added to the	18:003	0.01%	ida (in jargaren 19. auk	
STATE OF BE	13:047	0.01%	la comença e de de	
$p_{ij}^{a}(x) = \frac{a_{ij}^{a}(x)}{a_{ij}^{a}(x)}$	12.045	0.01%	true karconina	第17 7年 中国
1. 36.00	11.028	0'01%		
Ye gri.	2,928	0.00%		
at called the ways	864	0.00%		
50.86 367 15 597 636 F	780	0.00%	12,615	0.10%
(No. 11.75.2) (164)	328	0.00%		ASSAULT (ESTATE) Estantia de la composition della composition dell
Jedopolija	aravata da	0 00%	endered their section. The contract	Salar esta
Triedl	YV/3723,245	末肢 维特	to rectars (和原则数

- 4. There is no arbitrary class wide claim deadline. Deadlines are tied to the date of purchase and product warranties which in some cases can last between 15 and 30 years and for the original owner can last up to a lifetime.
- 5. The Settlement will be administrated by the Defendant through its warranty department. By tapping into the existing warranty relationship, the settlement claim process will follow the natural path a homeowner with a warranty issue would

- follow. This reduces the pressure on the notice plan to stay current for 15-30 years for those homeowners who may someday want to file a claim.
- 6. The Defendant's warranty program is one of the most extensive in the industry making it unlikely that homeowners would see any advantage in opting out to pursue their own action.
- 7. Products were distributed through building supply centres and big box retailers.

Target Audience

Persons who own or owned a building with a roof containing BP organic shingles manufactured between 1985 and 2010.

Household Characteristics

Although Canadian Households span a wide range of demographic indicators, the effectiveness of the media spend can be maximized by targeting the largest segments of each household characteristic.

Age of home owners: Ownership rates tend to rise with age, peaking at age 55-59 (75%), and declining very gradually thereafter. The lowest ownership rates are found in the 20-24 (20%) and 25-29 age range (42%).

Income of home owners: The median income of homeowners is \$60,000.

Location of home owners: 80% of Canadian homeowners live in urban areas. The proportion of Class Members living in urban areas will be slightly less, however, when high-rise condominium owners are removed.

Education: Education is correlated with income, and therefore, as income rises, education levels rise and home ownership rises.

Strategy

- 1. Target 25-65 year olds with income levels over \$40k using a combination of direct mail and paid media placements in print publications and television targeting. While we recognize there are home owners that fall outside this range, this demographic is representative of the majority of Canadian home owners.
- Using the defendant's historical sales data, break the target demographic into five key segments: Quebec, Ontario, Alberta, Rest of Canada (ROC), the United States, and concentrate media spending in those areas where the majority of sales took place (see Table II).
- 3. Utilize a combination of direct mail and paid media placements in print publications and television to find the appropriate balance between overall reach and cost. Consideration will be given to the extended claim period and the low probability for opt out's.

Table II - Plan Summary

Street Hitter	Baocini	Wanne	THE RESIDENCE OF THE PARTY OF T	WITHOUT SOURCE STUDIOS CALL	្រីស្រែងហេតុ	il development	
	35/36%	y y	V	ý	V.	V.	Ÿ
	28.70%			9	₩.		
	.10745%			V	ý	y 3	V
And white Property of the property of the company	1 10	75%			Car Section		(March
	4.84%	3 - V	Ver iver	1.0	cognic Way		(V.
	3.78%	i i	Continue	4		19-19 V 53-53	V
	3.66%		198-3 (1)	si Ve	100 av	V	i v
	3 21%	September 1	1000	11 - V	Procession Contract	i e A v asani	€/ × 3 V €
	2.85%	V	10 miles	4. 4. 4.	100 (100 (10 0 (100 (100 (100 (100 (100 (100 (100	9.	V
	2,63%	rai rasa w oo	P. Wind	ý.	a April N. April Sacratio	in the state of	12 W V
	(0)50%	in with	i yiri	3 3 V	Ŵ.	y,	V.
	0.02%	(P. (V)	of with the	Control of the Contro	hill grade for	9	¥,
	0.00%		V	V.	i i i i	W (2)	v v
8.99 (27)	** \$15.500 PM 100.0	21%	100 CAS	74470730444	13年15日的共和国	Barrie Bridge	E/15/4/20
	1:27%		10000000		Page 15 To Sept.	BRANCE A	
tan ang katawa. Nggangga Balatan at m	0.76%	S		1. 18 S	Service March	And Miller	
programme de la companya del companya del companya de la companya	0.54%		V	V 	V V	1	
nter ned Maria de la	0.46%		V	V	, v	19 W	
क्षात्र । प्रकास	0.38%		, v	0 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	N V	W.	
	0 21%	, X	N.	V.	X	v.	
	0.13%	N.		DECEMBER OF STREET	, v	v.	N
as do the self	0109%	1 3		V V		V	
	0.04%	ÿ		v	i i		N. C. C.
1-99 (4.1%)	0:03%	1 7			, v	V	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
A margania	0.02%		, n		i v	l V	\$15 CH
	0.01%	W	v l	W.	i v	i v	
	0.01%	v	v	Ű.	, v	í.	
prise Managorija	0.01%	, v	- N	y -	V.	y.	
	0.01%	l v	v :	v.		, W	5 X X
File	(0)00%	J	N.	N	v.	.	and the same of th
Called the couples	/ (0:00%	, v	, v	V.	y i		
	1 (0/00%	W.	W.	Ÿ.	, y	v i	ne de Str
Molecoped	0.00%	. v	W	i v	v 🐧	w.	AND MARKET
1996a (1907)	m (************************************	4%	As Carlo				250244

6. Review of Media Selection

1. Website

A bilingual website will be a key component of the Notice Plan. Its primary purpose will be to act as a repository of case information for class members. In addition to case documents such as the Settlement Agreement, it will have copies of the Short Form and Long Form Notice, the Claim Form, Opt Out Form and Frequently Asked Questions. It will also have contact information for questions or to order a Claim Package. The dedicated URL will figure prominently in the Notices and advertising with the intent of steering people toward the website rather than the call centre.

To ensure relevance and thus rankings within search engines, the site will be designed with many of the key search terms Class Members would be expected to use when searching for information on the case. Throughout the life of the website we will continue to monitor the rankings and optimize the site to ensure

relevance. Finally, the site map will be submitted to all search engines to ensure maximum exposure

2. Direct Mail

Phase I

A claim package consisting of a copy of the Long Form Notice, a Claim Form and an Opt Out Form will be mailed to Class Members on a list provided by the Defendant. This list will contain the names and addresses of individuals who have previously filed warranty claims with the Defendant that are not yet resolved. It is expected this list will contain approximately 5,000 names.

Phase II

A claim package consisting of a copy of the Long Form Notice, a Claim Form and an Opt Out Form will be mailed to Class Members who become aware of the Settlement and contact the Defendant requesting a Claim Package.

3. Press Release

A copy of the Short Form Notice in English and French will be issued through Marketwire. This will assist in getting the word out through credible news sources.

Canada:

This list includes all daily newspapers, all television stations and selected major radio stations in every city in Canada and US with a population over 50,000. It also includes every daily newspaper, the TV, and radio stations reached by The Canadian Press Wire Network across Canada, as well the major news agencies, broadcast networks, and media in six smaller provincial centers to offer the widest national reach.

United States:

This distribution puts the message in front of newspapers, broadcast outlets and relevant industry trade publications within the Northeast region. It also includes real-time, full-text delivery to Web sites and databases, to the AP and Dow Jones newswires, and to national publications such as The New York Times, Wall Street Journal and USA Today.

3. Trade Journals/Trade Associations

To supplement the print notice, and to specifically target roofing contractors and wholesalers/retailers, notice will be published in the Trade Journals listed in Table III.

Notice will also be forwarded to the Canadian Roofing Contractors Association and the National Roofing Contractors Association in the U.S. for distribution to their membership.

By targeting individuals directly tied to the roofing industry, these media provide high target composition at low cost.

Table III - Trade Journals

		, and the street	Mar Telephipa	parametric programment		
to Original Constitution of the Constitution o	i falli feakion-wak	105400	Marint (\$847) Marint (\$847)	(Green)	.£ 5.	
eschool 2 can be a character to be over a more	Bi-annually:	A.I	v	5 000	N/0	
straction as the major of the first	Nov & Mar	N	'	5,000	N/A	
edges form ha	Bi-monthly: Jan, Mar, May, Jul, Sep & Nov	N	Y	29,000	400	
Mill these later and	Monthly	Υ	Υ	N/A	25,000	
Business The Apple of	Monthly	N	Υ	N/A	83,616	
(secondario de la companya de la com	Monthly	N;	Y	N/A	44,000	
Most on action strain. Mathematics make basis situary Grand thing standy	Monthly	Υ	Υ	4,000	16,000	

3. Newspapers

Canada:

Use major market dailies in markets with population of 100,000+ and also Globe/Post which have national footprint. The recommendation is to use weekend insertions in all plus additional insertion on weekday in the Quebec, Ontario and Alberta markets that represent 75% of the historical sales.

In total, the selected newspapers have a circulation of over 17 million (weekday/Saturday) with a total readership of 37 million people (see Table IV – Canadian Newspapers)

Table IV - Canadian Newspapers

		Residentiño		theor	πi
1201/11/2017/11		Epinologyapa	:25-(totanel dance	805/001
States - 1997 Asperanga	Weekdey 948,000 Saturday 1,065,000	8534	25:54,69%	\$50,000 + \$75,000 + \$100,000 +	86% 66% 50%
	Weekday: 1,686,500 Saturday: 1,689,700	75%	25-64: 71%	<\$50,000 \$50,000 + \$75,000 +	30% 70% 46%
	Weekday 1.179.300 Saturday 1.549.900	76%	25/64 73%	\$50,000 ± \$75,000 ± \$100,000 +	73% 50% 34%
	Weekday: 395,600 Saturday: 567,900	63%	25-54: 57%	\$75,000+ \$100,000+	45% 28%
	Weekday 148 900 Saturday 205 000	(57%	85,54,87%	\$60,000 + \$75,000 +	57% 42%
	Weekday: 175,700 Saturday: 204,700	71%	25-64: 83%	\$30,000 + \$50,000 + \$100,000 +	20% 50% 15%
	Weekday:218,000 Saturday:268,000	34% (hömes greater ithan \$200k)	25 54 47%	\$100,000#	15%
	Weekday: 54,400 Saturday: 58,900	72%	Avg age: 53 yrs	Avg Income: \$56,300	The Control Military
	Weekday 45 400 Saturday 57 100	67%	Avg ege 50 yrs	Avglincome \$56,738	
ayındı ili kadrılığır	Weekday: 621,100 Saturday: 668,900	64%	18-49: 49%	\$50,000 + \$75,000 + \$100,000 +	52% 37% 21%
	Weekday, 226,200 Saturday, 245,800	60%	18-49:58%	\$50,000.+ \$75,000.+	25% 18%
	Weekday: 136,800 Saturday: 139,200	58%	18-50: 66%	\$75,000 +	68%
	Weekday 77 268 Seturday 86,721	70%	35:49:75%	(\$100,000+	78%

United States:

Given the small concentration of sales in these regions -4% of total defendant sales - the recommendation is to use one weekend insertion in dailies serving the communities identified in the Defendant sales data .

In total, the selected newspapers have a circulation of over 2.4 million (Saturday or Sunday) with a total readership of over 5 million people (see Table V – United States Newspapers)

Table V - United States Newspapers

Bullication	(fave)t-pp/d(ca)	Sinte	(0.37	Resoluting
Stage (strain b) the	BateVia*	/ NY	Sunday 1	12,667
Application in	Duke Center*	PA I	Saturday	11,200
Opening Color of the Color of the color	Plymouth	MA	Sunday	26,085
Bostone Spage	Buffalo, Cheekatowga, Lancaster, Amherst, West Seneca,	NY	Saturday	235,227
	Lake Placid, Tonawonda, East Concord			
directopy): 50 a fewsy	Burlington, Essex Junction, Milton, Colchester, Morrisville	VT	Sunday	40,482
sydnomic Amerikay	Concord	NH	Saturday	18,953
"Ophsein beite 19756.	Plymouth, Livonia, Detroit, Canton, Manchester*	Mil	Saturday	74,335
tyspectodapter. Aceth principes alripte	Andover	MA	Saturday	39,927
Appelyte course trapping probables	Bloomfield Hills Farmington	经接 M 图型	Sunday	147,880
districted priorities.	Hartford, Windsor	ст	Saturday	197,406
titus agadytis incitoopii yhtyiseksy kidanda	Lancaster	PA	Sunday	94,794
Butter be to retire to a Widelines a politic (Complie	Augusta	ME	Sunday	27,861
Wife and produced States	Moscow, Genesee*	3.4 .0 2.53	Sunday	24,563
currence Const. (currence)	Lewision	ME	Saturday	27,831
unttere Ade	Lexington	MA	Saturday	49,559
Approximate artists operated from a tangent appellation of	Manchester	NH	Sunday	62,394
A 97-33 who to Apopla 6.5 may	New Bedford, Dartmouth	MA	Sunday	25,526
That theoretic paths	Oxford	ст	Sunday	94,119
Yest strictly say	Weterford	(CT	Saturday	29,477
Margit Book gille the syndagorithm and the second of the second	Peru Chery	NY _e	-Sunday -	18,764
Martin and Archive Campaga Amplian Ampagaga and Ampagaga	Portland	ME	Sunday	76/715
Bateryophala (ugulfó	Portsmouth, Salem, Londonderry	NH	Sunday	13,681
Sadytares configurated design	Barrington, Providence	RI	Sunday	130,659
teach (tig): Mangage (tigo hours to	Rochester	NY	Sunday	175,146
Q(, mhoti, ' min) → folian	Scarnton	PA	Sunday	63,251
Augilie 1 marg	Seattle	A WAY	Sunday	345,491
appedicate in Appendic a president of the Appendicate in Appendica	Spokane Pullman	WA	Saturday.	90,461
\$97 types: Mysiksigtniftjeil	Syracuse Liverpool	NY	Sunday	140/225
Arilliansppg Camasperie	Middlebury Center	PA	Sunday	29,900
Attalia ayasar — aliabqirijin Bayayashara	Spencer, Lunenberg, Uxbridge	MA	Sunday	76,030
^e skings squite appelle	Grandview	WA	Sunday	33,237

4. Television

Close to 99% of Canadian and American households own a television. On average, people watch 26 hours of TV a week, which makes it one of the most effective media available.

We would recommend 100% focus on specialty networks. Although their reach is not as high as conventional (80% versus 95%), they are very cost efficient and have national footprint. To maximize reach we would suggest we purchase a number of stations (6-10) in both English and French Canada.

To minimize cost, we would recommend that we purchase run of station (ROS) whereby our spots would be distributed among all day parts. We may also consider a direct response strategy where we would buy remnant space – we would give the station some basic parameters (ie. make sure we have spots in all day-parts) but give them discretion to schedule our spots in unsold inventory. This would guarantee absolute lowest cost paid.

Sample Specialty Channels

উচ্ছেবালীলৈ	diamads
English	French
WiG-	HES:
History+	Series++
Spoisosile	Carrillos:
Discovery Ch	Z-tele+
ekcinya:	Gandidae
Space+	LCN+
Showers	R08:
Food Networ	VRAK+
Whatehers	Humas
Comedy+	ARTV#

7. Pricing

See attached pricing schedule

8. Measurements

The proposed plan will deliver the following national and regional reach:

National:

	Week i	Week2
All	65.1 %	77.9 %
TV spot	53.5 %	65.9 %
Newspaper daily	33.5 %	44.5 %
Online advertising	-	-

Ontario:

	Week1	Weekz
All	64.7 %	78.1 %
TV spot	52.2 %	65.4 %
Newspaper daily	34.4 %	45.7 %
Online advertising	-	-

Alberta:

	Weeki	Weeko
All	62.7 %	75.3 %
TV spot	50.8 %	64.0 %
Newspaper daily	32.4 %	40.9 %
Online advertising	-	-

Quebec:

	Welfis	Wiekż
Ail	60.6 %	72.1 %
TV spot	51.7 %	64.6 %
Newspaper daily	26.5 %	31.1 %
Online advertisin	9 -	-

BP CLASS ACTION SETTLEMENT OPT OUT FORM

INSTRUCTIONS: You do not have to take part in the BP class action settlement or be a member of the Class. This form can be used to exclude yourself (to "opt out") from the settlement and the Class. If you wish to remain a member of the Class and claim compensation under the settlement, DO NOT COMPLETE OR RETURN THIS FORM. Before deciding whether to remain in the Class or to request exclusion, make sure you have read the long-form settlement notice and understand the consequences of your decision. A copy of the long-form settlement notice is available online at [website] or by calling [number]. If you have any questions regarding the consequences of your decision, contact Class Counsel at the addresses listed below. If you exclude yourself, you cannot receive any payment from or object to the settlement. Any Court orders in this matter will not apply to you. By excluding yourself, you keep any right to file or proceed with a lawsuit about BP Organic Shingles that you may have.

To request exclusion, you must fully complete, sign, and return this form via First Class Mail postmarked no later than _____, 2012 to the applicable addresses below. If you do not return this form by this deadline, you will lose your right to exclude yourself and will be bound by this settlement and all subsequent proceedings, orders, and judgments entered by the Court in your jurisdiction. You cannot opt out without properly and timely completing this form.

Class Members with claims related to property(ies) in Canada (outside of the Province of Québec and those with buildings in the Province of Québec who employ more than 50 employees or were established in the public interest) should mail Opt-Out Forms to:

Jonathon Foreman Harrison Pensa LLP 450 Talbot Street P.O. Box 3237 London, ON N6A 4K3

For Class Members with claims related to property(ies) in the Province of Québec (except those who employ more than 50 people or were established in the public interest) should mail Opt-Out Forms to:

[name]

AND

Superior Court of Québec

Lauzon Bélanger Lespérance Inc. 286 Saint-Paul West, Suite 100 Montréal (Québec) H2Y 2A3 Palais de Justice 300, boul. Jean-Lesage Québec City, QC G1K 8K6

Class Members with claims related to property(ies) in the United States should mail Opt-Out Forms to:

US pl's counsel

AND

Vermont Superior Court

Chittenden Unit

175 Main Street Burlington, VT 05401

DAL01:1157214.4 1713377.2

BP CLASS ACTION SETTLEMENT OPT OUT FORM

Please print clearly in black or blue ink. Complete this form only if you wish to be excluded from participating in the BP class action lawsuit.

1. Your Name and Contact Information	
Name:	
Address:	
City: Province/State: Postale	
Are you completing this form on someone else's behalf?	YesNo
If Yes, list the name and address of the person for whom	you completing this form
Are you an attorney acting for a client?Yes	No
2. Property Information	
(1) Do you currently own property(ies) that may contain BP	Organic Shingles?
YesNo	
OR	
Did you previously own property(ies) that may contain B	P Organic Shingles?
YesNo	
(2) The property(ies) that contains BP Organic Shingles is lo	cated in (check all that apply):
P	rovince of QuebecUnited States
(3) In the space below, list the address(es) of the property(ies) that contains BP Organic Shingles.
(4) Specify the number of buildings or structures on each pro	perty that contains BP Organic Shingles.
(5) Specify the size of the buildings or structures upon that co	ontains BP Organic Shingles in square feet or square
meters.	
3. Legal Representation	
Have you entered into a written or oral agreement to be repre Shingles?	sented by counsel regarding your Br Organic
-	
YesNo	

If yes, li	st the attorney's name and	address if different from the person named i	n section 1 above:
Attorney	's Signature:		
BY ANY PENDIN TRUE: 1) 2)	JUDGMENT IN THE IG OR FUTURE LITION OR FUTURE LITION OF THE PROPERTY OF THE PR	MPLETE AND RETURN THIS OPT OU LITIGATION AND YOU WILL NOT BIGATION ON MATTERS RESOLVED In to the settlement; itigating a pending lawsuit regarding BP (stinue even if you do not opt out); and Out Form to an incorrect location or misson.	E PERMITTED TO PURSUE ANY N THE SETTLEMENT. THIS IS Organic Shingles (exception:
Dated:		Submitted by:	(Printed)
		Signed:	(Signature)
		Address:	

DAL01:1157214.4 1713377.2

BP ORGANIC SHINGLES CLASS ACTION SETTLEMENT CLAIM FORM

INSTRUCTIONS

How to obtain information about the BP class action settlement and determine your eligibility for filing a claim:

- Generally speaking, you will be eligible for compensation under the BP class action settlement if you meet the following criteria:
 - You are a class member. You are a class member if you own or owned a building in Canada or the United States whose roof contains or contained BP Organic Shingles. BP Organic Shingles means shingles made with a felt reinforcement base material that is saturated with asphalt, also known as "organic" roofing shingles, manufactured during the period from 1985 to 2010 under one of the following brand names: Eclipse, Eclipse H/R, Eclipse LS, Super Eclipse, Weather-Tite, Mirage, Rampart, Tradition, Tite-Lok, Esgard Pro-Standard, Pro-Standard, Esgard 20, Esgard 25, Citadel, Tite-On, Roofmaster, Roofmaster Classic, Roofmaster Plus, Elegance, Elegance II, Europa, and Super Lok.
 - You submit a timely and complete Claim Form, along with all other required information.
 - You have not already removed the BP Organic Shingles.
 - You have experienced qualifying damage to your BP Organic Shingles. Such damage includes blistering, clawing, cracking, curling, cold weather curling, fishmouthing, and delamanation. See section 4.5 of the settlement for complete information about the type of damage covered by the settlement.
 - The damage to the BP Organic Shingles was not caused by improper installation, improper maintenance, weather-related events, or some other factor unrelated to the manufacturing process. See section 4.6 of the settlement for complete information.
 - The applicable warranty period for your BP Organic Shingles has not yet expired.
- Please review the Settlement Notice for more information about the Settlement Agreement and whether you are eligible to file a claim. A copy of the Settlement Notice and Settlement Agreement is available online at [website] or by calling [number].

How to complete the Claim Form:

- To be eligible for settlement benefits, you must complete and submit this Claim Form.
- Please type or print all your responses in ink.
- All copies of documentation submitted in support of your Claim Form must be clear, legible and complete.
- Your claim number is: ______. Please identify your claim number on all documentation you send to BP in respect of your claim,
- Keep a copy of the Claim Form and all enclosed documents for your records. Materials submitted to BP will not be returned.
- A separate Claim Form must be filed for each building with BP Organic Shingles for which you are making a claim.
- You are required to complete all mandatory sections of the Claim Form. Mandatory sections are marked with an asterisk (*). The failure to complete all mandatory sections might result in the denial of your claim. Use "N/A" when the question does not apply.
- To the extent possible, you should complete all optional sections of the Claim Form. An incomplete Claim Form might result in a delay in the processing of your claim.

- BP might contact you for further information, documents or sample BP Organic Shingles. The failure to provide the requested information, documents or sample BP Organic Shingles might result in the denial of your claim.
- BP might contact you to schedule an inspection of your roof. The failure to cooperate with BP in scheduling an inspection might result in the denial of your claim.

Where to send your Claim Form:

Claim Forms must be sent to the following address:
 Building Products of Canada Corp.
 Claim no.: [insert your claim number]
 c/o: Warranty Services
 9510 St-Patrick Street
 LaSalle, Quebec
 H8R 1R9
 Fax:

Email:

How to determine the due date for your Claim Form:

- Claim Forms must be submitted no later than 150 days after the later of the Effective Date of the settlement and the Claimant's discovery of the need for repairs or replacement and, in any event, no later than the expiry of the applicable warranty period and prior to the removal of the BP Organic Shingles from the home, residence, building or other structure upon which they were installed. The applicable warranty period is available online at [website], listed in the Settlement Notice, and attached as a schedule to the Settlement Agreement. The Effective Date will likely be approximately [date-30 days after US hearing]. Once known, the Effective Date will be posted online at [website].
- Claim Forms postmarked after the applicable claims deadline cannot be processed. If the
 postmark is not legible, the Claim Form will be deemed to have been postmarked when actually
 received by BP.

What to expect after you submit your Claim Form:

- Please use a shipping method that provides delivery confirmation. You will not be provided with acknowledgement of receipt by BP.
- Be assured that your claim will be evaluated as promptly as possible, given the need for BP to
 properly and thoroughly evaluate each claim. In most circumstances, you should expect a
 response within 120 days of having provided BP with all the necessary information about your
 claim.
- You are responsible to provide all documents, shingle samples and information as requested by BP to permit claim processing. You must quickly respond to any request made by BP for additional information. If you fail to provide required information, your claim might be rejected.
- Once your claim has been evaluated, you will receive a written response by mail or by email (if an email address has been provided).

Please notify BP of any change of address that occurs after you submit your claim.

Claim Number:	
---------------	--

CLAIM FORM

To ensure that your claim is processed, please complete all mandatory fields of information.

Mandatory fields are marked with an asterisk (*)

Please write "N/A" if any question is not applicable

First Name*	÷	Last Name*			
Mailing Address*	City*	p	rovince/State*	Postal/	Zip Code*
Home Phone*	Work Phone	Cell Phone	Ema	il Address	
	ith BP Organic Shingles installe	······································	ent than Ivialling A	Address*	
Name of development,	neighbourhood or subdivision (if known):		•	No
Name of development, Has there been an assig		if known): your BP Organic	Shingles?*	Yes	No
Name of development, Has there been an assig If yes, you must pro	neighbourhood or subdivision (if known): your BP Organic with this Claim I	Shingles?*	Yes	No No
Name of development, Ias there been an assig If yes, you must pro- Iave you or a prior own hingles?	neighbourhood or subdivision (nment of the claims relating to y vide any related documentation	if known): your BP Organic with this Claim I BP in respect of	Shingles?*	Yes	
Name of development, Has there been an assig If yes, you must pro Have you or a prior own thingles? If yes, what was the	neighbourhood or subdivision (nment of the claims relating to y vide any related documentation ner previously filed a claim with approximate date of the claim? me of the claim? (including	if known): your BP Organic with this Claim I BP in respect of	Shingles?*	Yes	

BUILDINGONNERSER			
Do you currently own the building?	Yes	No	
Does anyone else currently own an interest in the building	? Yes	No	
If yes, you must complete the section below entitled "C	o-Owner".		
When did you acquire the building? (yy/mm/dd)			
When was the building constructed? (yy/mm/dd)			
If you currently own the building, please provide two form	ns of proof of	f ownership:*	

a copy of
3

First Name*		Last Name*		
Home Phone*	Work Phone	Cell Phone	Ema	ail Address

Brand name of shingles and colour:* You must attach proof of purchase and installation of BP Organic Shingles. The proof of purchase must establish that the shingles were BP Organic Shingles and the date that the shingles were installed.* Proof of purchase might include an invoice from a roofer for services already rendered or a receipt from the building supply store where the shingles were purchased. Quotes, bids or estimates are not acceptable. If the BP Organic Shingles are the original installation, please provide a certificate of occupancy, a final building inspection or an invoice from the builder or roofing contractor showing the date of installation.* Certificate of Occupancy Provided Yes N/A Final Building Inspection Provided Yes N/A

Invoice from the builder or roofing contractor showing the date of installation Provided

Yes N/A

Please attach any previous correspondence you have received from BP confirming that the shingles on the structure are BP Organic Shingles.

placulitiedibility	outhrol antoquerd	ntiethyousehin		* <u>************************************</u>	
ontractor Name			Contact		
failing Address		City		Province/State	Postal/Zip Code
Vork Phone	– Fax		Cell Phone	T	il Address

DD ADDUR Olip La galler pardion y d	i apropi ce di Più	Plegalicora pomenta	llangsinglyjaallaad	gliodos:	
Dealer Name			ontact		
Mailing Address		City	Province/St	ate Postal/Zip Code	
Work Phone	Fax		Cell Phone	Email Address	

ROOT SHRUCTURE

Please indicate the type of structure upon which you have experienced damage to your BP Organic Shingles. Please check each that apply:

Single Dwelling Residential Home

Duplex or Multi-Residential Home

Garage (Heated)

Garage (Unheated)

Commercial Property	
Total Number of Slopes of your Roof:	
Roof Slope (if known) (ie 4/12):	
Please provide the following information regarding the measurement of your roo	of:*
Measurements of your roof on a slope-by-slope basis:*	Square Feet Square Metres
	Square Feet Square Metres
The number of roof bundles or squares installed on the roof. A "bundle" is a package of shingles. A "square" means 100 square feet of roofing shingles. A "square" is typically composed of three bundles/packages of shingles.	Bundles Squares
Where the number of roof squares is not available, you must provide the followin	g information:*
Square footage of the roof structure:	
Square footage of the ground floor of the building:	
Number of stories of the building:	
Please provide all other information in your possession that might assist BP in the measurement of your roof; *	determination of the
Please attach a drawing of your roof, showing all roof slopes.	
O(0)) ((0)) (GDPN) (Sept. 1) (Sept.	organice) biologica (* 1772) 18 de januario (* 1886)
What do you see?*	
Where is this happening on your roof?*	

Is the roof leaking?	Yes	No	: '	,		
If yes, where is it leaking?						
Is there visible ice accumulation on your roof?	Yes	No				
Number of roof slopes with dama	ged BP C	Organic Shin	gles*:			
Number of damaged BP Organic S multiple roof slopes have damage number of damaged BP Organic S	d BP Org	anic Shingle	s, piease sp	ecify the		
	ion from	an insurer,	ouilder, Shingles?*	Yes	No	

PHIOTOGRAVITAS

You must attach photographs that establish the condition of the BP Organic Shingles. The photographs must be labelled with your name and claim number, and identify the slope or area of the roof shown. Photographs can be submitted in hard copy or electronically. Photographs must be submitted in colour.

Please provide any report that you have received from a roofing contractor documenting the condition of your BP

Organic Shingles, including any damage to the BP Organic Shingles.

The photographs must show:*

the structure from a distance, sufficient to show the entire structure including views of all roof vents

the inside and outside of all soffits

at minimum, two photographs of each roof slope showing the slope and the condition of the shingles

SIDTOTO PEAVADNOT BYONG TOTOTS

You must select a settlement benefit option. For more information about the settlement benefits options, please see Notes of the Settlement Notice.*

Cash Settlement Option. Under the cash settlement option, during the non-proration period, eligible Class Members in Canada will receive Cdn \$75/square and eligible Class Members in the United States will receive US \$75/square. After the non-proration period, the Cdn \$75/square (or US \$75/square, as appropriate) will be prorated or discounted to account for the use the eligible Class Members have already received from their BP Organic Shingles.

Repair Settlement Option. Under the repair settlement option, eligible Class Members will be compensated for the costs of repairing and replacing the damaged BP Organic Shingles, in accordance with the terms of the applicable warranty. During the non-proration period, BP will reimburse the eligible Class Member for the full costs of repairs or replacement in accordance with the terms of the applicable warranty. After the non-proration period, BP will reimburse the eligible Class Member for the costs of repairs or replacement in accordance with the terms of the applicable warranty on a prorated basis, to account for the use the eligible Class Members have already received from their BP Organic Shingles. In completing the repairs or replacement, eligible Class Members can select the roofer of their choice, but must use shingles manufactured by BP.

Copies of the applicable warranties, including details of the non-proration periods, are available online at [website].

ACCANOMADIDACITMITIMITY

Please check the boxes below to confirm your acknowledgement of the following:

Claimants must acknowledge that they have read and agree to the following by checking the boxes (mandatory):*

VERIFICATION OF CLAIM AND WARRANTY. Claimant(s) represents and warrants that the information, enclosures and supporting documentation submitted herewith are true, correct, and accurate. Claimant(s) specifically warrants that Claimant(s) is the rightful and only owner or assignee(s) of the Claim submitted and has not otherwise transferred or encumbered any right or interest in this Claim and/or right or entitlement arising from the BP class action settlement to any person.

AGREEMENT TO COOPERATE. Claimant(s) hereby agrees to cooperate with BP in providing any additional information, documents or photographs necessary for the determination of this Claim,

Claimant(s) further agrees to comply with any request by BP to provide one full size shingle from the structure that is the subject of this Claim. The shingle sample must be representative of the condition of shingles on the relevant roof slope. If such a request is made, BP will reimburse Claimants in Canada up to Cdn \$50.00 and Claimants in the United States up to US \$50.00 for the expense of having a roofing professional remove, replace and ship the shingle sample. Such reimbursement will be made provided that: (1) you provide invoices establishing expenses incurred; (2) the shingle, upon inspection and testing, is determined to be a BP Organic Shingle; and (3) you are determined to be a Settlement Class Member.

If you are required to send additional information or a shingle sample to BP, you should ensure tracking of documents and samples you are sending as there will be no acknowledgement of receipt.

AUTHORIZATION OF INSPECTION. Claimant(s) hereby agrees to allow BP and the Inspector (who was approved by the Courts to review disputed claims decisions) access to the property that is the subject of this claims, including the interior of the structure, for the purposes of evaluating this claim.

If an inspection is required, please indicate whether you would like to be present for the inspection:	Yes	No
Please indicate whether the property has any features (locked gates, security system, aggressive dog, etc.) that might interfere with an inspection:		

RELEASE. In consideration of the benefits provided by the BP class action settlement, and subject to the terms of the BP class action settlement, Claimant(s) fully and finally settle, release and discharge BP, and its predecessors, successors, present or former parents or subsidiaries, affiliates, related parties, officers, directors, employees, agents, attorneys, representatives, suppliers, distributors, vendors and assigns, which for greater certainty includes, without limitation, Emco Limited, Emco Corporation and BPCO Corp. (collectively, the "Released Parties"), from each and every claim of liability, including relief under federal law or the law of any province, territory or state, which arises out of Damage to BP Organic Shingles, including without limitation all claims or liability on account of or related to Damage to BP Organic Shingles, which were alleged or could have been alleged in the relevant class action and all claims for penalties, punitive damages, exemplary damages, statutory damages, damages based upon a multiplication of compensatory damages, court costs, or attorneys' fees or expenses, which might otherwise have been made in connection with any claim relating to Damaged BP Organic Shingles. Without limiting the generality of the foregoing, the release relates to: (1) any claim for breach or violation of any federal or provincial/state statute, consumer protection acts, case law, common law, or other law; (2) any claim for breach of any duty imposed by law, by contract or otherwise; (3) any claim based on strict product liability, failure to warn, negligent failure to warn, negligence, reliance, breach of express or implied warranty, racketeering, fraud, conspiracy, consumer fraud, negligent misrepresentation/omission, or intentional misrepresentation/omission; (4) any claim arising from or in any way related to the design, manufacture, production, labeling, warning, distribution, handling, written or oral product instructions, specification, marketing or use of the BP Organic Shingles, and/or any alleged defects in the BP Organic Shingles, or any part thereof; and (5) any claim for penalties, punitive damages, exemplary damages, or any claim for damages based upon any multiplication or enhancement of compensatory damages associated with (1) through (4) above.

Please see a copy of the Settlement Agreement for full details of the release. A copy of the Settlement Agreement is available online at [website] or by calling [number].

SUBMISSION TO JURISDICTION OF COURTS. Claimant(s) resident in Canada (excluding Claimants resident in Quebec, save and except entities located in Quebec with 50 or more employees or which were established in public interest who are members of the Canadian national class) agrees to submit to the exclusive jurisdiction of the Ontario Superior Court of Justice for all purposes associated with this Claim. Claimant(s) resident in Quebec agrees to submit to the exclusive jurisdiction of the Quebec Superior Court for all purposes associated with this Claim. Claimant(s) resident in the United States agrees to submit to the exclusive jurisdiction of the United States Court for all purposes associated with this Claim.

Claimant(s) further agrees to advise any subsequent purchaser of the property that is the subject of this claim of the release and make such other appropriate disclosure as may be required by applicable local, provincial and state laws regarding the purchase and sale of the property.

Claimant(s) hereby warrants and represents that Claimant(s) has not assigned or transferred or purported to assign or transfer, voluntarily or involuntarily, any matter released pursuant to this release or any other part or portion thereof.

[NTD: Mike, do we need to ask for tax information for US residents?]

WE NAMES AND ASSESSED THAT HE AND ASSESSED AND ASSESSED AND ASSESSED ASSESSED ASSESSED AND ASSESSED AS

I/we certify that all the information and documentation that I/we have provided to BP is true and accurate to the best of my/our knowledge and belief. All photographs that I/we have provided are typical of the damage to the BP Organic Shingles installed on the roof and/or roof slopes for which compensation is sought. This document is signed under penalty of perjury.

If more than one owner, the Claim Form must be signed by	,
Signature of Claimant	Date (yy/mm/dd)
<u> </u>	
Signature of Co-Owner	Date (yy/mm/dd)

The Claim Form must be sent to the following address:

Building Products of Canada Corp. Claim no.: [insert your claim number] c/o: Warranty Services 9510 St-Patrick Street LaSalle, Quebec H8R 1R9 Fax:

Email:

Court File No.: 4367/11CP

ONTARIO SUPERIOR COURT OF JUSTICE

Proceeding commenced at LONDON.

ORDER

HARRISON PENSA LLP London, ON N6A 4K3 Barristers & Solicitors 450 Talbot Street

Genevieve Meisenheimer (LSUC #59191U)
Tel: (519) 679-9660
Fax: (519) 667-3362 Jonathan J. Foreman (LSUC #45087H)

Lawyers for the Plaintiffs

File No: 148924